



This is the peer reviewed version of the following article:

López Puertas-Lamy, M., Desender, K., & Epure, M. (2017). Corporate social responsibility and the assessment by auditors of the risk of material misstatement. *Journal of Business Finance & Accounting*, 44 (9-10), pp. 1276-1314

which has been published in final form at https://doi.org/10.1111/jbfa.12268

This article may be used for noncommercial purposes in accordance with Wiley Terms and Conditions for Use of Self-Archived Versions.

Corporate Social Responsibility and the Auditors' Assessment of the Risk of Material Misstatement

Mónica López Puertas-Lamy* Department of Business Economics; Universidad Carlos III Calle de Madrid, 126, E-28903 Getafe (Madrid), Spain. mlopezpu@emp.uc3m.es

Kurt Desender Department of Business Economics; Universidad Carlos III Calle de Madrid, 126, E-28903 Getafe (Madrid), Spain. kdesende@emp.uc3m.es

Mircea Epure Department of Economics and Business Universitat Pompeu Fabra, Barcelona GSE and Barcelona School of Management Ramon Trias Fargas, 25-27 E-08005 Barcelona, Spain mircea.epure@upf.edu

* Corresponding author: Mónica López Puertas-Lamy

Abstract

This paper investigates whether, and how, firms' Corporate Social Responsibility (CSR) Performance influences the auditors' assessment of the risk of material misstatement, whether due to fraud or error, at the financial statement level by analysing their pricing decision (i.e., audit fees). Using a panel data set of 12,330 firms from 28 countries over the period 2003-2012 and different measures of CSR performance, we find a U-shaped relationship between firms' CSR performance and audit fees. This result suggests that there is an optimal level of CSR performance that minimizes the auditors' assessment of the risk of material misstatement, which in turn lowers the need for greater auditor effort; that is why auditors charge firms significantly less when their CSR performance is at the optimal level. Finally, we also show that the optimal level of CSR performance varies with the degree of environmental dynamism, ownership concentration and leverage.

Keywords: Corporate Social Responsibility (CSR); CSR performance; Audit fees; Risk of material misstatement.

1. Introduction

The past two decades have witnessed a dramatic increase in the firms' engagement in Corporate Social Responsibility (CSR) in response to the needs and expectations of a wide range of stakeholders (Campbell, 2007; Waddock, 2008; Hoepner et al., 2016).¹ Meanwhile, numerous information intermediaries have emerged, rating firms across several dimensions of environmental and social performance, to provide credible CSR ratings and scores in a standardized and comparable way. The increasing interest in CSR, as well as the availability of CSR scores, has led to a proliferation of academic studies seeking to better understand its determinants and consequences (Huang and Watson, 2015).

Much of the existing literature on the consequences of CSR has focused on analysing the impact of CSR on different measures of firm performance (e.g., Margolis et al., 2007; Di Giuli and Kostovetsky, 2014; Eccles et al., 2014; Flammer, 2015; Lys et al., 2015), on firm value (e.g., Renneboog et al., 2008; Gregory and Whittaker, 2013; Gregory et al., 2016), on access to finance (e.g., El Ghoul et al., 2011; Goss and Roberts, 2011; Cheng et al., 2014; Hoepner et al., 2016), and on (post-audit) financial reporting quality² (e.g., Petrovits 2006; Chih et al., 2007; Prior et al., 2008; Kim et al., 2012), with the general conclusion that CSR affects all these dimensions but with mixed results. However, to the best of our knowledge, the relationship between CSR performance and the auditors' assessment of the risk of material misstatement, whether due to fraud or errors, at the financial statement level (i.e., the auditors' assessment of the risk that the financial report contains material misstatements before the

¹ Increasingly, customers, investors, employees and other stakeholders are considering the firm's CSR in their decisions. To illustrate, a large number of institutional investors are signatories to the United Nations' "Principles for Responsible Investment" (UNPRI). The most recent UNPRI website reports 1718 investment institutions as signatories, representing 59 trillion of assets under management. Moreover, the socially responsible investing (SRI) movement has gained importance and professionally managed US assets tied to SRI, accounted for more than 3 trillion dollars in 2010 (Di Giuli and Kostovetsky, 2014). In addition, Nielsen's (2014, 2015) Global Survey on Corporate Social Responsibility shows that 67 percent of the 30,000 participant in 60 countries prefers to work for socially responsible companies, while 60 percent indicate that they are willing to pay more for products and services from companies that are committed to positive social and environmental actions (up from 50% in 2013). ² The existing literature on the link between CSR and financial reporting quality (Petrovits, 2006; Chih et al., 2007; Prior et al., 2008; Kim et al., 2012) analyses the link between CSR and earning quality constructs that are computed based on financial reports that are issued, not before but, after the completion of the audit. We therefore refer to this stream of literature as linking CSR to (post-audit) financial reporting quality.

audit is conducted, hereafter, the auditors' assessment of the risk of material misstatement³) is still far from known, and has received little attention from academics and policy makers.⁴

The current paper addresses this issue and investigates whether, and how, a firm's CSR performance influences the auditors' assessment of the risk of material misstatement, by analysing their pricing decision (i.e., audit fees). With this aim, we follow Mackey et al. (2007) and the European commission (2001) and consider CSR practices as the voluntary firm actions designed to improve social or environmental conditions. This concept of CSR corresponds to what Baron (2006) denominates Corporate Social Performance (or CSR performance).

Understanding the implications of CSR performance on the auditors' assessment of the risk of material misstatement is important for academics and policy makers, since this assessment affects the auditors' effort and evaluation of evidence (Hammersley et al., 2010) and ultimately the (post-audit) firms' financial reporting quality. In this sense, theory predicts that higher audit effort increases the likelihood of detected errors and reduces the likelihood of undetected errors at the financial statement level (Shibano, 1990; Matsumura and Tucker, 1992; Dye, 1993; Hribar et al., 2014), implying a positive relation between the audit effort and (post-audit) financial reporting quality. Empirically, Lobo and Zhao (2013) provide support for this prediction, finding a robust negative association between audit effort (measured by audit fees) and annual report restatements. Therefore our analysis may help to understand whether the audit effort is one mechanism through which CSR performance influences the (post-audit) financial reporting quality of firms and hence, whether it is an important omitted variable of prior studies that examine the effect of CSR performance on (post-audit) financial reporting quality (Petrovits, 2006; Chih et al., 2007; Prior et al., 2008; Kim et al., 2012).

³ Misstatements in the financial statements can arise from either fraud or error. In this paper, we focus on both types of misstatements. The distinguishing factor between fraud and error is whether the underlying action that results in the misstatement of the financial statements is intentional or unintentional. The International Standards on Auditing (ISAs), establish that two types of intentional misstatements are relevant to the auditor: misstatements resulting from fraudulent financial reporting and misstatements resulting from misappropriation of assets (ISA 240). Given the relevance of fraud, greater responses and procedures should be implemented when the auditor believes that there is risk of material misstatement due to fraud rather than to error (ISA 240).

⁴ Anecdotal evidence reiterates the increasing role of CSR information in auditing. For example, the KPMG managing director, Eric Israel, noted at the 2010 Amsterdam Global Conference on Sustainability and Transparency that they have begun to better understand the implication of this information for their audits but that more work is needed. However, as far as we know, there is no empirical evidence on whether and how CSR performance influences the auditor's assessment of the risk of material misstatement and hence the audit fees.

We extract the auditors' assessment of the risk of material misstatement from the audit fees since they reflect the amount of effort that the auditors are expected to expend in the audit to fulfil its purpose (Srinidhi and Gul, 2007; Rice and Weber, 2012; Ghosh and Tang, 2015) and auditors respond to increases in misstatement risk by increasing their audit efforts (ISAs 330). The purpose of an audit is to obtain reasonable assurance on whether the financial statements as a whole are free from material misstatement. To obtain reasonable assurance, the auditor should gain sufficient appropriate audit evidence to reduce the risk of issuing an incorrect opinion on the financial statements (i.e., audit risk) to an acceptably low level (ISA 200). This audit risk is a function of the risk of material misstatement and the detection risk such that:

Audit Risk = Risk of material misstatement in the financial statements x Detection Risk

The risk of material misstatement is the entity's risk and is outside the direct control of the auditor (it exists independently of the audit of the financial statements). In contrast, the detection risk is under the control of the auditor since it constitutes the risk that the procedures performed by the auditor to reduce the audit risk to an acceptably low level will not detect a misstatement that exists and that could be material either individually or when aggregated with other misstatements (ISA 200).

According to the audit risk model, auditors choose an acceptable level of audit risk and then assess the risk of material misstatement, which leads to the desirable level of detection risk (Houston et al., 1999). When the auditor's assessment of the risk of material misstatement is high, the detection risk needs to be held at a lower level to keep the audit risk at an acceptable level. Lower detection risk can be achieved by increasing the nature, timing and extent of the audit procedures. These actions in turn increase the cost of the audit, as they require more effort and/or the involvement of personnel with more overall or industry-specific experience⁵ (Bedard and Biggs, 1991; Solomon et al., 1999; Johnstone and Bedard, 2003). Thus, auditors charge higher audit fees on engagements in which they assess the risk of material misstatement to be high. Conversely, when the auditor believes that the risk of material

⁵ Professional guidance suggests that the risk of material misstatement should lead the auditor to plan more hours (ISA 330). In addition, the positive linkage between the risk of material misstatement and audit work has been documented in prior studies (e.g., Hackenbrack and Knechel, 1997; Bell et al., 2001; Johnstone and Bedard, 2001).

misstatement is low, the detection risk can be set at a relatively higher level, reducing the audit effort or the specific experience of the engagement team and, hence, the audit fees (ISA 200).⁶

Auditors assess the risk of material misstatement by gaining an understanding of the entity and its environment, including the entity's internal control (ISA 315) and the incentives/pressures, opportunities, and attitudes/rationalizations that managers or employees may have to commit fraud (ISA 240). In doing so, the auditor obtains, among other things, an understanding of the entity's objectives and strategies, and of those related business risks⁷ that may result in risks of material misstatement. The business risk is broader than the risk of material misstatement, though it includes the latter since most business risks will eventually have financial consequences and, therefore, an effect on the financial statements (ISA 315). For example, the business risk arising from a contracting customer base may increase the risk of material misstatement associated with the valuation of receivables or may increase manager's incentives/pressures to engage in fraudulent financial reporting. Thus, an increase in the firm's business risk should lead to an increase in the auditor's assessment of the risks of material misstatement. In this line, O'Keefe et al. (1994); Pratt and Stice (1994), Johnstone (2000), Lyon and Maher (2005) show that audit partners recommend higher audit effort and fees in response to higher business risk.

Building on resource dependence and agency theory, we claim that within certain limits, an increase in the firm's CSR performance is likely to reduce the auditor's assessment of the risk of material misstatement, and hence the audit fees, by reducing the firm's business risk. As CSR performance continues to increase, however, this positive effect of CSR performance on the reduction of the auditor's assessment of the risk of material misstatement is likely to level off and eventually turn into a negative effect because of increasing business risk and auditor's concerns related to the opportunistic use of CSR. Therefore, we posit that the CSR performance - audit fee relationship may

⁶ We restrict our sample to firms with Big-4 auditors. Similar to Ghosh and Tang (2014), we assume similar quality among the Big-4 auditors (i.e., we assume that they choose the same level of audit risk). Thus, for a given level of audit risk, higher levels of audit fees should reflect higher assessment of the risk of material misstatement by the auditor.

⁷ We adopt the definition of business risk provided by ISA 315 which defines it as "a risk resulting from significant conditions, events, circumstances, actions or inactions that could adversely affect an entity's ability to achieve its objectives and execute its strategies, or from the setting of inappropriate objectives and strategies".

be described as a U-shape, such that there is an optimal level of CSR performance that minimizes the assessed risk of material misstatement and, as a result, the audit fees.

The general outline of our argument is as follows. For relatively low levels of CSR performance, an increase in the firm's CSR performance is likely to reduce the firm's business risk, and as a result the auditor's assessment of the risk of material misstatement, and hence the audit fees by a) enabling the firm to secure and retain critical resources controlled by various stakeholders (Hillman and Keim, 2001; Wang et al., 2008), and b) improving the firm's foresight capacity (Orlitzky et al., 2003), allowing managers to better anticipate the future and act timely to avoid negative consequences. As CSR performance continues to increase, however, this reduction effect of CSR performance on the firm's business risk will level off, and once a certain level of CSR performance is achieved, a further increase in the firm's CSR performance will eventually increase the auditor's assessment of the risk of material misstatement and the audit fees by increasing a) the firm's business risk (due to constraints on stakeholder support, direct costs, increasing risk of failing to deliver upon CSR promises, and stakeholders' concerns about the existence of agency problems within the firm); and b) the auditor's concerns about the opportunistic use of CSR.

Moreover, we posit that environmental dynamism, ownership concentration and leverage are likely to moderate the relationship between CSR performance and audit fees by influencing the auditors' assessment of the risk of material misstatement associated with a given level of CSR performance. On the one hand, in a dynamic environment, a firm is likely to be more dependent on its stakeholders for critical resources (Berman et al., 1999; Frooman, 1999) and to have a lower ability to accurately assess the impact of any environmental change on the firm (Milliken, 1987). We claim that in such situations, CSR performance becomes more important to reduce the firm's business risk and that, as a result, the degree of environmental dynamism may influence the CSR performance-audit fees relationship, increasing the level of CSR performance that minimizes the audit fees. On the other hand, the firm's degree of ownership concentration and leverage may also affect the relationship between CSR performance and audit fees, by influencing the auditor's concerns about agency problems linked to CSR performance. Consistent with Grossman and Hart (1980), Stulz (1990), Diamond (1991) and Shleifer and Vishny (1997), we conjecture that when firms have a high degree of ownership concentration or leverage, the ability of managers to employ CSR performance for non-value maximizing reasons is limited (Barnea and Rubin, 2010), since owners and creditors play a more active monitoring role. In addition, the required payments under debt contracts also reduce the available funds that could be employed for non-value-maximizing CSR performance (Jensen, 1986). As a result, the level of CSR performance that minimizes the audit fees is likely to increase as the firm's ownership concentration and/or leverage become higher, since auditors may be less concerned about the opportunistic use of CSR.

To test our conjectures, we draw on a comprehensive international sample of listed firms from the most relevant stock market indices over the period 2003-2012. Similarly to Ioannou and Serafeim (2012) and Lys et al. (2015), our data on CSR comes from ASSET4, a Thomson Reuters database that provides objective, relevant and systematic scores on firms' environmental, social, economic, and governance performance, based on over 250 objective key performance indicators (KPIs) and over 750 individual data points. To measure CSR performance, we combine the social and environmental performance scores from ASSET4. Because audit quality and client characteristics vary between Big-4 and non-Big-4 audits (e.g., Craswell et al., 1995; Behn et al., 2008), we restrict our sample to firms with Big-4 auditors, similarly to Ghosh and Tang (2015), to avoid introducing any bias from variations in the sample composition.

Our results show a robust quadratic relationship between CSR performance and audit fees, so that there exist a level of CSR performance associated with minimal audit fees. In particular, the regression estimates indicate that audit fees are at a minimum level when the CSR performance score (which moves between 0 and 1) is equal to 0.35. Below (beyond) this point, audit fees decrease (increase), in an economic and statistically significant way, for each marginal increase in CSR performance increases for firms in dynamic environments and for firms with higher degrees of ownership concentration or leverage.

We conduct extensive tests to ensure the robustness of our results. First, to ensure that our results are not an artefact of size, which is one of the main drivers of audit fees (Hay et al., 2006), or of the relatively large proportion of US-firms in our sample, we re-estimate the audit fees regressions for different subsamples. In particular we split our sample between a) firms with smaller and larger than median yearly industry sales and b) US and non-US listed firms. We find consistent results across all subsamples. Second, we show that our results are not driven by a single dimension of our CSR performance measure as they hold for the social and environmental dimensions that make up this measure. Third, we show that our results are robust to using the D-SOCIAL-KLD measure of CSR performance developed by Carroll et al. (2016) on a subsample of US firms.⁸ Finally, we estimate a proxy of the pre-audit misstatement risk as in Lobo and Zhao (2013) on a subsample of US firms, to provide additional support to our conjecture that there is an optimal level of CSR performance that minimizes the auditor's assessment of the risk of material misstatement. Our results provide support for this conjecture, showing a quadratic relationship between CSR performance and the pre-audit misstatement risk.

This paper contributes to the literature in several ways. First, we contribute to research that looks into factors that influence the auditor's assessment of the risk of material misstatement (e.g., Dechow et al, 1996; Bell and Carcello, 2000; Shelton et al., 2001; Bedard and Johnstone, 2004) and to the related literature on understanding the characteristics of firms with a higher risk of restatements (e.g., Dechow et al., 2010). To the best of our knowledge, this is the first study to explore the link between CSR performance and audit fees, and to propose a U-shaped relationship between these two variables. Bedard and Johnstone (2004) argue that one important challenge that audit firms face is to identify the riskiest clients from their portfolios and to plan and price those engagements. Our results suggest that auditors take CSR performance into account in their assessment of the risk of material misstatement such that there exists an optimal level of CSR performance that minimizes the audit fees. This finding may assist in the development of client acceptance and retention guidelines and could help to refine the traditional models of audit fees.

⁸ The D-SOCIAL-KLD measure is only available for US firms.

Second, we contribute to the literature on the relationship between CSR and (post-audit) financial reporting quality. Within this literature, two main results have emerged. On the one hand, Kim et al. (2012) find that CSR firms are less likely to engage in aggressive earnings management and in real activities manipulation and that their CEOs/CFOs are less likely to be the subject of SEC investigations of GAAP violations, consistent with ethical concerns driving managers to produce highquality financial reports. On the other hand, Petrovits (2006) and Prior et al. (2008) report a negative relationship between CSR and (post-audit) financial reporting quality which is interpreted within the agency theory framework as the result of the opportunistic use of CSR. In particular, Petrovits (2006) find evidence of the strategic use of corporate philanthropy programs to achieve earnings targets, while Prior et al. (2008) report a positive link between CSR performance and earnings management and claim that managers may use CSR strategically to disguise earnings management. Finally, Chih et al. (2007) examine the link between CSR and earnings management but report conflicting results. Our study may help to further understand the mixed results found in this literature. Given the documented positive relation between audit fees and (post-audit) financial reporting quality (e.g., Shibano, 1990; Matsumura and Tucker, 1992; Dye, 1993; Hillegeist, 1999; Lobo and Zhao, 2013), our results points to audit effort as a mechanism through which CSR performance may influence the (post-audit) firm's financial reporting quality. In particular, our findings suggest that audit fees may be an important omitted variable for prior studies that examine the relationship between CSR performance and (post-audit) financial reporting quality. When analysing the relationship between CSR performance and (post-audit) financial reporting quality without controlling for audit fees, it may therefore not be possible to determine whether CSR performance increases the financial reporting quality directly, or whether the relationship works through increased audit effort. By examining the relationship between CSR performance and the auditor's assessment of the risk that the financial report contains material misstatements before the audit is conducted our analysis helps to provide a better understanding of whether CSR performance is linked to more transparent and reliable financial information provided by the firm (i.e., the pre-audit financial reporting quality).

Third, we argue and show that environmental dynamism, ownership concentration and leverage have a significant moderating effect on the CSR performance-audit fees relationship, which suggests that a firm's operating environment and corporate governance plays an important role in the effect that CSR performance has on the auditor's assessment of the risk of material misstatement. These findings respond to the call for the consideration of contingencies when studying CSR performance (McWilliams and Siegel, 2001; Barnett, 2007).

Finally, we employ a large international sample and present robustness findings across different subsamples and for different measures of CSR performance, providing external validity to our findings.

The remainder of the papers is structured as follows: section 2 develops our theoretical framework. Section 3 outlines our research design while section 4 describes our data and sample. Section 5 presents the results of our analyses. Section 6 reports the sensitivity analyses. Finally, Section 7 presents our conclusions and suggestions for further research.

2. Theoretical framework

In this section, we develop the arguments that the relationship between CSR performance and the auditor's assessment of the risk of material misstatement, and hence the audit fees, is captured by a U-shape. We first revise the literature on the factors that influence the auditor's assessment of the risk of material misstatement. Then, we establish how CSR performance is likely to influence these factors, differentiating between positive and negative implications of CSR performance for the reduction of the auditor's assessment of the risk of material misstatement. Finally, we analyse the likely trends of these positive and negative implications, to propose a U-shaped relationship between CSR performance and audit fees.

2.1 Factors influencing the auditor's assessment of the risk of material misstatement.

When planning the audit, the auditor has the responsibility to identify and assess the risks of material misstatement through understanding the entity and its environment (ISA 315), including the entity's internal control and the incentives/pressures, opportunities, and attitudes/rationalizations that

managers or employees may have to commit fraud⁹, thereby obtaining a basis for designing and implementing responses to the assessed risks of material misstatement.

One important aspect for the auditor to consider when assessing the risk of material misstatement is the firm's business risk since most business risks will eventually have financial consequences and therefore an effect on the financial statements (ISA 315). To illustrate, the business risk resulting from troubled customer relations may increase the risk that provisions for impairment of relevant accounts receivables are not made adequately. In addition, an increase in the firm's business risk may be considered by the auditor as a fraud risk factor as it may result in an incentive or pressure to engage in fraudulent financial reporting practices (ISA 240). For instance, if the financial stability or profitability is threatened by economic, industry, or entity operating conditions (such as significant declines in customer demand or flaws in a product or service that may result in liabilities and in a loss of reputation), managers or employees may face pressures or have incentives to commit fraudulent financial reporting through, for instance, inappropriate revenue recognition.¹⁰

A considerable stream of research (e.g., Colbert, 1988; Kizirian et al., 2005; Johnstone et al., 2014; Ghosh and Tang, 2015) has identified different factors that influence the firm's business risks and hence the auditor's assessment of the risk of material misstatement. This research suggests that the environment in which an organization operates (e.g., the state of the economy), the characteristics of its employees (e.g., ethical values, experience and turnover), the quality of information systems, the internal audit function, the management integrity and operating style as well as the financial position

⁹ Research on fraud risk and ISA 240 explains fraud as the interaction of three causal influences affecting a potential perpetrator: incentives/pressures, opportunities, and attitudes/rationalizations. Incentive results from a perceive pressure on a person to commit fraud. For example, management may be given large bonuses that are based on accounting income. Opportunity results from conditions that allow a person to commit fraud. For example, lack of appropriate monitoring may allow management to inappropriately manipulate reported earnings to reach analyst's forecasts. Attitude is what allows a person to rationalize the act of fraud. For instance, manager's unethical behaviour. Importantly, each of these factors is a necessary but not a sufficient condition for fraud, and as such, fraud risk assessment should consider the interaction of these factors (Wilks and Zimbelman, 2004). As established by the ISAs, auditors should perform the fraud risk assessment when planning the audit to modify the nature timing and extent of their audit procedures in case they perceive either incentives/pressures, opportunities, and attitudes/ rationalizations to commit financial fraud.

¹⁰ For instance, Bristol-Myers Squibb agreed to pay \$150 million to settle fraud charges in 2004. The SEC alleged, among other things, that Bristol-Myers engaged in a fraudulent scheme to inflate its sales and earnings in order to meet its internal sales and earnings targets and analysts' earnings estimates.

and accounting practices of the organization (e.g., poor financial position and inadequate working capital) are important factors that affect the auditor's assessment of the risk of material misstatement.

We claim that the firm's CSR performance also influences the firm's business risk and the auditor's assessment of the risk of material misstatement. In elaborating this argument, we rely on resource dependence theory because it focuses on the influence of stakeholders on the firm's decision-making and its consequences (Pfeffer and Salancik, 1978). This theory suggests that because the allocation of the resources necessary for a firm's survival are often not fully controlled by the firm, but rather by key stakeholders, the firm faces uncertainty in securing and retaining those resources (Pfeffer and Salancik, 1978; Frooman, 1999). Thus, the auditor, in accessing the risk of material misstatement, may be concerned by the way the firm addresses and manages its stakeholder relations, being CSR one key mechanism to manage these relations. In what follows, we comment on how the firm's CSR performance may reduce or increase the auditor's assessment of the risk of material misstatement.

2.2 Positive implications of CSR performance for the reduction of the auditor's assessment of the risk material misstatement.

We argue that CSR performance is likely to reduce the firm's business risk, and as a result the auditor's assessment of the risk of material misstatement, by a) enabling the firm to secure critical resources controlled by stakeholders, b) reducing the risk of losing resources it already controls and c) enhancing the firm's forecast ability.

First, as CSR performance explicitly considers the interests of various stakeholders into the firm's business model and operations, it may allow firms to expand the set of value-creating exchanges with its stakeholders and, as a result, to ensure critical resources controlled by the firm's key stakeholders, including current and prospective employees, customers, shareholders, regulators, and the community (Hillman and Keim, 2001; Wang et al., 2008).

According to the social identity theory, CSR has the potential to increase the work morale of employees and their efficiency (Edmans, 2011; Schmitz and Schrader, 2015) and to attract and retain high quality employees (Greening and Turban, 2000; Jones and Murrel, 2001). As a consequence,

employees of firms that engage in CSR may show greater commitment, higher ethical values, education or experience than employees of firms that do not engage in CSR. Higher moral standards, abilities and commitment of employees are likely to reduce their incentives and/or attitudes to engage in fraudulent financial reporting or in misappropriation of assets that may lead to misstatements. In this line, Gao et al. (2014) show that executives of CSR firms profit significantly less from insider trades and are less likely to trade prior to future news than executives of non-CSR firms. Therefore, since the reliability of the financial reporting process depends in good part, on the employees and management's integrity and ability to identify and discuss important financial reporting issues¹¹ (Cohen and Hanno, 2000), the firm's CSR performance may reduce the auditor's assessment of the risk of material misstatement, whether due to errors or fraud.

CSR performance may also help the firm to attract socially conscious consumers, which are generally willing to recompense CSR firms by increasing their demand for their products or services, by accepting higher prices or by showing greater loyalty (Waddock and Graves, 1997; Hillman and Keim, 2001; McWilliams and Siegel, 2001; Baron, 2008). CSR performance could thus be an instrument to gain sustainable competitive advantage (Porter and Kramer, 2006; Gregory et al., 2016; Hoepner et al., 2016) and to prevent consumer boycotts (Glazer et al., 2008), reducing the business risk, and as a result, the auditor's assessment of the risk of material misstatement.

Similarly, investors and creditors may be more willing to invest in firms with high levels of CSR performance (Graves and Waddock, 1994; Clarkson et al. 2004; Barnett and Salomon, 2006). Thus, CSR performance may reduce the firm's cost of financing and enhance it access to finance (Mackey et al., 2007; Cheng et al., 2014; Ioannou and Serafeim, 2015; Plumlee et al., 2015; Hoepner et al., 2016). This is evidenced by the growth of socially responsible investing (Di Giuli and Kostovetsky, 2014). Better investor relations through CSR performance may help to alleviate short term pressure on financial performance and to reduce the incentives to engage in fraudulent financial reporting or in misappropriation of assets that may lead to misstatements.

¹¹ A study by KPMG (1999) on incidents of corporate fraud found that a majority of the cases were related to factors present in the control environment, such as a lack of management integrity.

Finally, CSR performance may also reduce the auditor's assessment of the risk of material misstatement by mitigating the likelihood of negative regulatory, legislative or fiscal actions (Freeman, 1984; Berman et al., 1999; Maxwell et al., 2000; Hillman and Keim, 2001), since regulators may be more supportive of CSR firms. In this spirit, Maxwell et al. (2000) present an adjustment model in which a reduction of emissions below the measure required by law lowers the probability of stricter regulations in the future. Thus, according to their model a firm can apply CSR activities to prevent a sharper regulation and hence to reduce the firm's business risk.

Second, CSR performance can also reduce the firm's business risk and hence the auditor's assessment of the risk of material misstatement by helping the firm to reduce the risk of losing resources it already controls (Brammer and Millington, 2004; Godfrey, 2005; Barnett and Salomon, 2006). In this sense, different studies have indicated that CSR performance a) should be regarded as a form of reputation building or maintenance (Fombrun and Shanley, 1990; McWilliams et al., 2006; Freeman et al., 2007; Gregory et al., 2016; Hoepner et al., 2016), b) produces insurance-like effects on the firm's stock and bond price (Godfrey, 2005; Godfrey et al., 2009; Peloza, 2009; Minor and Morgan, 2011; Shiu and Yang, 2017) and c) helps poorly performing firms to recover from a disadvantageous position more quickly (Choi and Wang, 2009; Vanhamme and Grobben, 2009).

Third, research has shown that CSR performance improves the firm's foresight capacity (Waddock, 2002; Orlitzky et al., 2003), enabling the firm to better anticipate the future and to act in a timely manner regarding external changes and turbulence to avoid negative consequences. Increased competency in anticipating changes in the business environment may reduce the firm's business risk, improve management planning and decrease the risk of misstatements in inventory, accounts receivable, and other accounts, reducing the auditor's assessment of the risk of material misstatement.

2.3 Negative implications of CSR performance for the reduction of the auditor's assessment of the risk material misstatement.

Although managerial discretion in CSR may enable managers to reduce the firm's business risk, it is equally reasonable to suspect that if there exists a principal-agent problem within the firm, for

example when the owners cannot perfectly control the behaviour of the manager, managers may use CSR performance as a means to advance their personal interests (Friedman, 1970) at the expense of the firm's owners and other stakeholders (Jensen and Meckling, 1976; Carroll, 1979; McWilliams et al., 2006; Surroca and Tribo, 2008). Specifically, managers may use CSR to increase their own reputation and awareness level within social circles, to further political or career agendas, as part of an entrenchment strategy (Surroca and Tribo, 2008), or to create an impression of transparency and ethical behaviour among stakeholders to cover up corporate misconduct (Hemingway and Maclagan, 2004) or earnings management (Prior et al., 2008). These opportunistic motivations of managers to engage in CSR might lead to excessively high levels of CSR performance from the owner's point of view and may create additional incentives for earnings manipulation. In this line, the literature has shown that opportunistic managers are more likely to mask non-optimal expenditures by accounting manipulation (De Angelo, 1988; Christie and Zimmerman, 1994; Warfield et al., 1995; Gul and Tsui, 1997). Thus, when CSR performance is the result of agency problems¹², it is likely to increase a firm's business and fraud risk unless managerial misconduct can be constrained. In addition, CSR performance may also be the result of the existence of slack resources. In this situation, CSR performance is also likely to increase a firm's business and fraud risk since the existence of slack resources increases the risk of agency problems (Jensen and Meckling, 1976). Thus, if stakeholders perceive the firm's CSR performance as the result of managerial opportunism or as a signal of the existence of a large pool of slack resources, they may respond by withholding resources and support, leading to a further increase in the firm's business and fraud risk.

Therefore, CSR performance may increase the auditor's' assessment of the risk of material misstatement if auditors believe (or if they think that stakeholders believe) that a) managers act in their own interest when engaging in CSR at the expense of the firm's owners and other stakeholders, or b)

 $^{^{12}}$ In this spirit, Cheng et al. (2013) find that that spending on CSR is partly related to agency problems. In their analysis they find a decline in CSR performance when manager's ownership of firms increases. They conclude that "some forms of goodness investment, not on the margin, may increase firm value; however, managers spend the marginal dollar on goodness because they wish to do good with other people's money". Similarly, Atkinson and Galaskiewicz (1988), focusing on one aspect of CSR – namely, charitable giving, show that firms with high levels of CEO ownership give less generously to charities than firms with low levels of CEO ownership. They interpret this finding as evidence of better alignment between CEO and shareholder incentives when CEO ownership is higher.

the firm's CSR performance is a signal of the existence of slack resources. Consistent with this idea, Gul and Tsui (1997) show that auditors under the presence of agency problems within the firm increase their assessment of the risk of material misstatement and, as a result, they exert greater effort and charge higher audit fees.

CSR performance may also increase the auditor's concerns about the competitiveness of the firms since achieving higher levels of CSR performance imposes increasing direct cost on the firm, such as the costs resulting from offering employee day care or paid parental leave, reducing the carbon footprint or donating to charity (Barnett and Salomon, 2006). In addition, as the CSR performance increases, there is likely to be an increase in the administrative and human resource costs, as firms may need to create specialist CSR departments and CSR initiatives generally require, at least some, dedication on the part of employees (McWilliams and Siegel, 2001; Brammer and Millington, 2004; Wang et al., 2008). These additional costs directly detract from the firm profitability, and so may place firms with high levels of CSR performance at a competitive disadvantage, relative to those with lower levels (Friedman, 1970; Jensen, 2001). Thus, the direct cost of CSR performance may affect the firm's ability to achieve its objectives and as a result increase its business and fraud risk. In this situation, manager may have greater incentives /pressure to commit fraudulent financial reporting through, for instance, inappropriate revenue recognition.

Finally, failure to adequately take care of negative environmental and social issues, while portraying a high level of CSR performance is likely to lead to reputational losses and to greater exposure to future operational risk which is likely to create incentives for manipulation. Thus, CSR performance may increase the auditor's concerns about the firm's risk of failing to deliver upon the CSR promises and, as a result, the assessment of the risk of material misstatement.¹³

¹³ For example, shares of oil giant BP plunged after its failure to stop the oil leak in the Gulf of Mexico in April 2010, eliminating more than \$23 billion in market value in one day—and more than \$67 billion in less than two month after the disaster began (Reuters, June 2, 2010). The shareholders' reaction, triggered by the inadequate handling of the leakages, provides some anecdotal evidence on the link between CSR practices and financial risk. In addition, the SEC alleges that BP made fraudulent public statements, underestimating the magnitude of the oil spill. BP agreed to settle the SEC's charges by paying \$525 million. Similarly, Volkswagen's historical commitment to CSR was strongly questioned after the emissions scandal came to light (Forbes, Sept. 27 2015).

2.4 The U-shaped relationship between CSR performance and audit fees

We propose that, up to a certain level of CSR performance, the positive implications of CSR performance for the reduction of the auditor's assessment of the risk of material misstatement outweigh the negative implications, such that an increase in CSR performance reduces the audit fees. However, as the CSR performance continues to increase, the positive implications start to level off while the negative implications become stronger, up to a level of CSR performance where the negative implications outweigh the positive implications such that an increase in CSR performance leads to an increase in the firm's audit fees. In Figure 1, we plot the likely trends of the positive and negative implications of CSR performance for the reduction of the auditor's assessment of the risk of material misstatement and their effect on the firm's audit fees, at various levels of CSR performance.

[Figure 1 about here]

As shown by curve A, the positive implications of CSR performance for the reduction of the auditor's assessment of the risk of material misstatement, are expected to reduce the audit fees at first because of positive stakeholder responses, insurance-like effects, as well as the development of competencies in dealing with external changes. However, the reduction of audit fees due to the positive implications of CSR performance (i.e., the slope of curve A) is expected to level off as CSR performance continues to increase for two main reasons.

First, despite stakeholders' greater willingness to provide critical resources to firms that engage in CSR, there are limits to the resources that socially inclined stakeholders are able and willing to invest in the firm. This puts a natural constraint on the reduction of the business risk, and hence on the reduction of the auditor's assessment of the risk of material misstatement and audit fees that a firm can obtain from CSR performance. For example, the ability to attract and retain better employees, and to foster ethical behaviour is unlikely to grow at a constant rate as CSR performance increases. In this line, Wright and Ferris (1997) provide evidence of a diminishing marginal effect of CSR on attracting investments, such that too much CSR displaces investments, regardless of investor's preferences. Similarly, Flammer (2015) shows that environmental CSR is a resource with decreasing marginal returns. Second, as firms increase their CSR performance, they will inevitably transfer part of the cost to achieve a higher CSR performance to its stakeholders through, for example, lower wages, higher prices, or lower financial returns (McWilliams and Siegel, 2001). While CSR conscious stakeholders may be prepared to earn a lower wage, to pay a higher price, or to receive a lower financial return to support the firm's CSR performance, they however expect reasonable returns from their interaction with the firm, especially compared to alternative options (e.g., other firms with acceptable CSR performance levels). Thus, as CSR performance continues to increase, these stakeholders must, at some point, become reluctant to accept the terms of their interaction with the firm, and, as a result, start to reduce or withdraw their specific investments in the firm, increasing the firm's business risk and the auditor's assessment of the risk of material misstatement.

Regarding the auditor's concerns about the direct cost of CSR performance, represented by curve B, we believe that these concerns are likely to lead to an increase in the audit fees as CSR performance increases, but at a decreasing rate. Since there are important initial costs to set up CSR initiatives and to get employees and other stakeholders involved, the direct cost of CSR performance may at first increase the firm's business risk and, consequently, the auditor's assessment of the risk of material misstatement and the audit fees. However, due to learning effects and potential economies of scale linked to CSR performance, we expect the auditor's concerns about the direct cost of CSR performance increases.

In terms of the auditor's concerns about the opportunistic use of CSR performance represented in curve C, we believe that these concerns are likely to be relatively small at low levels of CSR performance. However, as CSR performance increases, the auditor's concerns about the potential opportunistic use of CSR by managers are likely to become larger, leading to an increase in the audit fees. This may be so because a certain level of CSR performance is needed to raise suspicion about a dishonest use of CSR or about the existence of slack resources. We expect the auditor's concerns about the opportunistic use of CSR to eventually level off as CSR performance continues to increase because severe misbehaviour related to CSR is likely to be constrained by corporate governance mechanisms. Finally, the risk of failure to adequately take care of negative environmental and social issues, is more important as CSR performance becomes higher. Thus, we expect the auditor's concerns about the firm's risk of failing to deliver upon the CSR promises represented in curve D, to lead to an increase of audit fees only at high levels of CSR performance.

Taking these opposing forces together, a U-shaped relationship between CSR performance and the audit fees emerges (curve E).

2.5. Contingency factors

McWilliams and Siegel (2001) and Barnett (2007) call for adopting a contingency perspective when studying CSR to gain a better understanding of the underlying variables and processes involved. Building on our conceptual framework, we propose three contingency factors that are likely to influence the relationship between CSR performance and the auditor's assessment of the risk of material misstatement. In particular, we consider that, on the one hand, the degree of environmental dynamism is likely to enhance the positive implications of CSR performance for the reduction of the auditor's assessment of the risk of material misstatement by increasing the firm's dependence on its stakeholders for critical resources. On the other hand, we argue that the firm's degree of ownership concentration and leverage are likely to reduce the negative implications of CSR performance for the reduction of the auditor's assessment of the risk of material misstatement by reducing the agency concerns related to CSR performance. Therefore, we expect the optimal level of CSR performance to be higher in dynamic environments or in firms with a high degree of ownership concentration or leverage.

Environmental dynamism

Environmental dynamism refers to the degree and the instability of changes in a firm's competitive environment (Dess and Beard, 1984; Boyd, 1990). Dynamic environments are characterized by greater environmental uncertainty, rapid changes in industry structure, the instability of market demand, and high probability of external shocks (e.g., Jansen et al., 2006; Levinthal and Myatt, 1994; Sirmon et al., 2007). We claim that in dynamic environments, the positive implications of

CSR performance for the reduction of the auditor's assessment of the risk of material misstatement are likely to be enhanced, for the following two reasons:

First, an increase in the degree of environmental dynamism reduces the firm's ability to a) accurately assess the impact of any environmental change on the firm and b) to determine the viable alternatives which managers can pursue and the potential impact of decision-making on current and future business activities (Milliken, 1987). In such situations, the positive implications of CSR performance for the reduction of the firm's business risk may be enhanced since stakeholder relations and higher foresight capacity become more important to ensure the firm's survival (Berman et al., 1999; Frooman, 1999). In this sense, Glazer et al. (2008) argue that in markets with higher product market competition, more CSR activities are to be expected, while Fernández-Kranz and Santaló (2010) show that product market competition is positively associated with CSR.

Second, a more dynamic environment increases the likelihood of unexpected events which may negatively impact some stakeholders. In the occurrence of such events, the firm risks losing the stakeholders' resource commitment and support (Godfrey 2005), which could leads to an increase in the firm's business risk, thus increasing the likelihood of errors in different accounts and the incentives for engaging in financial fraud. The firm's CSR performance may help to reduce the negative consequences of unexpected events by reducing the firm's risk of losing resources it already controls.

In contrast, the effect of environmental dynamism on the negative implications of CSR performance for the reduction of the auditor's assessment of the risk of material misstatement may not be significant. In particular, auditor's concerns that stem from the firm's failure to adequately address negative environmental and social issues or from the direct cost of CSR performance are less likely to shift because of environmental dynamism, while the concerns about agency problems may move upwards or downwards, making the net effect unclear. In dynamic environments managers may have greater ability to use CSR opportunistically as they require greater discretion in their decision-making (Galbraith, 1973). However, high environmental uncertainty may also require managers to allocate additional resources to deal with external shocks, which reduces the amount of slack resources (Farrell, 2001) that could be used opportunistically. In addition, in dynamic environments the opportunistic use

of CSR may have a stronger negative impact on stakeholder commitment, firm survival and the ability to achieve shareholder objectives. As a result, environmental dynamism may act as a deterrent of the opportunistic use of CSR. Taken together, these arguments suggest that environmental dynamism can influence the auditor's concerns about the opportunistic use of CSR in opposite directions and, as a result, we assume that they do not change in a significant way.

Thus, for all these reasons, we suggest that in a dynamic environment the level of CSR performance that minimizes the audit fees will be higher than in a less dynamic environment.

Ownership concentration and leverage

We propose that the capital structure of the firm, in terms of the degree of ownership concentration and leverage may influence the negative implications of CSR performance for the reduction of the auditor's assessment of the risk of material misstatement by influencing the auditor's agency concerns associated with CSR performance.

Regarding the firm's ownership structure, Berle and Means (1932) argue that in publicly traded firms, diffuse ownership structures reduce shareholders' incentives to monitor managers' wasteful expenditures. In contrast, large stockholders can reduce manager-shareholder agency conflicts, because they have powerful incentives to monitor managers (Grossman and Hart, 1980; Shleifer and Vishny, 1997; Ke et al., 1999). McConnell and Servaes (1990) provides support for these claims as they demonstrate that shareholders who own large stakes in a firm are more effective monitors. Effective monitoring may reduce the risk of opportunistic use of CSR. In this line, the ISA 240 establishes that "ownership characteristics of the entity have a significant influence on the consideration of relevant fraud risk factors" by auditors, while Dechow et al. (1996) note that the likelihood of earnings manipulation is systematically related to poor governance structures and weaknesses in management oversight. Thus, high levels of ownership concentration may reduce the auditor's concerns about the opportunistic use of CSR since large owners may be better informed and have greater power and incentives to restrict managers from undertaking CSR for opportunistic reasons.

Similarly the debt market may also provide management discipline (Rubin, 1990). Since high debt levels may induce creditors to play a more active monitoring role (Diamond, 1991; Gilson, 1990),

and the required payments under debt contracts reduce the funds available for non-value-maximizing activities, high debt levels may reduce managers' ability to use CSR performance as a means to advance their personal interests. Therefore, high debt levels may also reduce the auditor's concerns about the opportunistic use of CSR.

In contrast, neither the auditor's concerns related to the direct cost or to the risk that the firm may fail to adequately address negative environmental and social issues, nor the positive implications of CSR performance for the reduction of the auditor's assessment of the risk of material misstatement are likely to be affected by the degree of ownership concentration and leverage.

Thus, taken together these arguments, we propose that as the ownership concentration and/or leverage increase, the level of CSR performance that minimizes the audit fees will also increase.

3. Research design

An extensive body of literature examines the level and nature of audit fees in organizations. Most existing research on audit fees is based on the seminal work by Simunic (1980), who was the first to develop a positive model on the determinants of audit fees. To test the relationship between CSR performance and audit fees, we first use the model of Simunic (1980), as follows:

Audit fees_{it} = $\alpha + \beta_1 CSR$ -Score_{it} + $\beta_2 CSR$ -Score²_{it} + $\beta_3 Size_{it} + \beta_4 Return-on-Assets_{it} + \beta_5 Loss-in-current$ $year_{it} + <math>\beta_6 Current$ -Ratio_{it} + $\beta_7 Receivables$ -and-Inventory-to-Total-Assets_{it} + $\beta_8 H$ -index_{it} + $\beta_9 Auditor$ -Tenure_{it} + $\beta_{10} Auditor$ -Specialization_{it} + Country Dummies + Industry Dummies + Year Dummies + $\varepsilon_{it}(1)$

Consistent with the audit literature, we define¹⁴ our dependent variable (*Audit fees*) as the natural logarithm of the total fees charged by the auditor for the financial statements audit work. The independent variables of interest are *CSR-Score* and its quadratic term (*CSR-Score*²). We are mainly interested in the coefficient of these two variables (β_1 and β_2). If, as proposed, there is an optimal level of CSR performance associated with minimal audit fees, β_1 should be negative while β_2 should be positive. Similarly to Ioannou and Serafeim (2012), Lys et al. (2015) and Luo et al. (2015), our measure

¹⁴ Complete definitions of all variables are presented in Appendix A.

of CSR performance combines the social and environmental scores obtained from the Thomson Reuters ASSET4 database.

The control variables related to client characteristics are defined as follows. Size is the natural logarithm of total assets. Size is one of the most important determinants of audit fees, as it generally accounts for a large proportion of the variation in audit fees (Hay et al., 2006). Larger firms tend to have more accounts and more complex transactions, and thus larger audit fees. Return-on-assets is the ratio of net income to total assets. Higher return-on-assets ratios are expected to reflect lower financial risk and therefore a negative association with audit fees is expected. Loss-in-current-year is an indicator variable equal to one when the current year's net income is negative, and zero otherwise. Reporting a loss in the current year indicates higher financial risk and is expected to be positively related to audit fees. Current-ratio is the ratio of current assets to current liabilities. This liquidity measure is expected to show a negative relationship with audit fees. Receivables-and-inventory-to-total-assets is the sum of account receivables and inventory to total assets. We expect a positive relation between this variable and audit fees since receivables and inventories have been identified as two of the most difficult areas to audit. *H-index* is the sum, per industry-year, of the squared ratio of firm sales to total industry sales. Higher values of this market concentration index are associated with lower firm failure risk. Therefore, we expect to find a negative relationship between this variable and audit fees. The control variables related to auditor characteristic are auditor-tenure and auditor-specialization. Auditor-tenure is defined as the number of years of the audit engagement.¹⁵ Following the mixed evidence in previous studies (Hay et al., 2006), we have no clear prediction on the direction of this variable. Auditor-specialization is the auditor's market share of the client's industry, defined as the sum of the total assets of all clients in an industry divided by the total assets of all firms in that industry during the year. We expect a positive sign for the coefficient of auditor specialization, as specialized personnel may be more costly (Bell et al., 2001). In addition, due to the characteristics of our international sample, we add country, industry and year dummies.

¹⁵ We use the time span 1998-2012 (starting five years previous to our overall sample) to calculate auditor tenure. Thus, the maximum value of this variable is higher than the number of analyzed years.

Since Simunic's seminal study on the pricing of audit services, the audit pricing literature has further developed (e.g., Gregory and Collier, 1996; Ghosh and Lustgarten, 2006; Hogan and Wilkins, 2008), enhancing the list of relevant control variables. To reduce omitted correlated variables concerns, we use Hay et al. (2006) and Hogan and Wilkins (2008) to specify the following enhanced audit fee pricing model:

Audit fees_{it} = $\beta_1 CSR$ -Score_{it} + $\beta_2 CSR$ -Score²_{it} + $\beta_3 Size_{it}$ + $\beta_4 Return$ -on-Assets_{it} + $\beta_5 Loss$ -in-currentyear_{it} + $\beta_6 Current$ -Ratio_{it} + $\beta_7 Receivables$ -and-Inventory-to-Total-Assets_{it} + $\beta_8 H$ -index_{it} + $\beta_9 Auditor$ -Tenure_{it} + $\beta_{10} Auditor$ -Specialization_{it} + $\beta_{11} Leverage$ + $\beta_{12} Current$ -Assets-to-Total-Assets + $\beta_{13} Sales$ -Growth + $\beta_{14} Market$ -To-Book-Value + $\beta_{15} Extraordinary$ -Items + $\beta_{16} Busy$ -Season + $\beta_{17} Auditor$ -Change + $\beta_{18} IFRS$ + $\beta_{19} US$ -GAAP + Country dummies + Industry dummies + Year dummies + ε_{it} (2)

With respect to equation (1), in equation (2) we add additional client firm and auditor characteristics. Variables measuring client characteristics are as follows. Leverage is the ratio of the sum of the long-term and short-term debt to total assets. We expect a positive sign of leverage on audit fees since leverage is linked to the risk of a client failing, which potentially increases the risk of material misstatement at the financial statement level (Hay et al., 2006). Current-Assets-to-Total-Assets is the ratio of current assets to total assets. Sales-growth is the percentage growth in sales. We expect positive signs for both of these variables' coefficients (β_{12} and β_{13}) since firms with large amounts of current assets and high growth ratios may require additional audit attention. We also add the Market-to-Bookvalue, which is the ratio of the sum of the market value of common equity, the book values of preferred stock, and the book value of total debt to the book value of total assets. Extraordinary-Items is an indicator variable equal to one when a firm reports discontinued operations and extraordinary items and zero otherwise. A positive relationship is expected for the last two variables and audit fees as they capture elements of firm complexity. The variables measuring auditor characteristics are as follows: Busy-Season is an indicator variable equal to one when client's fiscal year-end is December or January and zero otherwise (with the exception of Japan, where it is March and April). Auditors are known to have a busy season close to the end of the fiscal year, which could lead to higher staff costs. Also, they could offer discounts for audit service outside the busy season. Auditor-Change is an indicator variable equal to one when the client engages a new auditor in a given year and zero otherwise. Changing auditors is usually related to lower audit fees (Butterworth and Houghton, 1995; Hay et al., 2006), which

may be the result of discounts being offered to attract new business. Finally, we introduce two dummies that take the value of one for IFRS and US GAAP reporting standards, respectively. DeFond and Zhang (2014) review the recent literature and find that the increased procedure required by IFRS could lead to higher audit fees, compared to local GAAP. Similarly, the adoption of US GAAP by non-US firms is likely to increase the audit fees, because of the complexity related to being listed in the US.

4. Data and sample description

For our empirical investigation, we start from all the firms for which there is detailed audit fee data and CSR performance information available in the Thomson Reuters ASSET4 database over the period 2003-2012. The CSR information provided by this database includes, for each company, performance scores for four pillars: environmental, social, economic and governance, as well as an overall performance score. All these scores range between zero and one. These scores are based on over 250 objective key performance indicators (KPIs) and over 750 individual data points. Our measure of CSR performance is the *CSR-Score* variable (hereafter CSR-Score) which we obtained from combining the social and environmental performance scores from ASSET4, assigning equal weights to both scores.¹⁶ Environmental performance refers to the firm's resources reduction, emission reduction, and product innovation benefiting the environment. Social performance captures how well the firm handles issues related to product responsibility, community, human rights, diversity, training and development, health and safety, and employment quality (see Appendix B for a description of how ASSET4 builds its performance scores and of the different categories within the social and environmental pillars).

Similar to Hoepner et al. (2016), we verify that our results are not driven by the construction of our measure of CSR performance re-estimating our model using the D-SOCIAL-KLD measure of CSR performance developed by Carroll et al. (2016). This measure is restricted to US listed firms and builds on, and improves, the Kinder, Lydenberg, Domini and Co. (KLD)¹⁷ measure of CSR performance by

¹⁶ Our CSR performance measure does not reflect economic performance or corporate governance performance, because those are less connected with the notion of social investments (Lys et al., 2015).

¹⁷ KLD compiles annual ratings of over 3,000 publicly traded U.S. firms, and its rates companies on a wide range of activities that reflect how well companies perform in social responsibility and build relationships with various

weighting "hard" CSR activities more than "easy" CSR activities (Carroll et al., 2016). Carroll et al. (2016) claim, and show, that their approach produces a measure of CSR performance that offers a more reliable comparison of firms than the standard KLD measure since adding up observable traits to calculate the KLD score is likely to produce measurement errors. Their approach is similar to the approach used by ASSET4 which defines relative levels of importance for each KPI.

We merge our initial dataset with firm-level stock market data from Datastream and accounting information from Worldscope. Detailed controls at the auditor level are obtained from Thomson ONE. We restrict our sample to firms that have a Big-4 auditor for consistency (only about ten percent of the initial observations had a non-Big-4 auditor). Finally, we winsorize our main variables at the one percent top and bottom level to avoid outlier-related problems. Our final sample comprises 12,330 firm-year observations from 28 countries over the period 2003-2012.

[Table 1 about here]

Table 1 presents the summary statistics of the main variables used in the audit fees regressions. The mean (median) audit fees are \$4.3 (\$3.47) million, while the mean CSR-Score is 0.57 and ranges from 0.06 to 0.98. We observe very similar statistics for the Social-Score and the Environmental-Score. In addition, the average D-SOCIAL-KLD score is 3.4 and ranges between -4.8 and 11.9. In terms of the contingency factors, environmental dynamism has a mean of 0.009, while the mean stake held by all blockholders amounts to 25 percent and the mean value of leverage is 55 percent.

Regarding the control variables, the mean total assets are \$2,692 billion, while the median value is \$675 billion. The average return on assets is 5.60 percent, while just above 10 percent of firms report a negative net income in our sample. The ratio of receivables and inventory over total assets is 0.27, while current assets are on average 1.75 times larger than the current liabilities. In terms of market concentration, the H-index is on average 0.19. The mean auditor tenure in our sample is around 8.7 years and the auditor specialization is on average 0.25. With respect to the additional control variables we add to the Simunic model, it is worthwhile highlighting that the mean ratio of current assets to total

stakeholders. KLD captures over 94 measurement items along seven social dimensions: product safety, diversity, employee relations, community relations, corporate governance, environmental stewardship, and human rights.

assets is 44 percent. In addition, sales growth is 8.7 percent on average and the market-to-book ratio is on average 2.92. In terms of the additional auditor characteristics, auditor change is relatively infrequent as less than 4 percent of firms change auditor in any given year. Finally, about 73 percent of firms in our sample have their fiscal year end during the auditors' busy season.

[Table 2 about here]

Table 2 presents the (pairwise) correlation matrix between the main variables of interest. The correlation between the CSR-Score and audit fees is positive with a level of 0.25, which is indicative of a significant overall positive relationship in the absence of relevant controls. The D-SOCIAL-KLD is also highly correlated with audit fees, the CSR-Score and the Social and Environmental scores. In addition, as established in the auditing literature, we observe a strong positive correlation between size and audit fees. Furthermore, audit fees show relatively high correlations with return-on-assets, current-ratio and the proportion of receivables and inventory to total assets, which highlights the importance of properly controlling for size and other firm characteristics.

5. Empirical results

5.1. CSR performance and audit fees

In Table 3, models (1) and (2) estimate the Simunic model with country, industry and year dummies (equation 1). In model (1) we include the CSR-Score, while in model 2 we also include its quadratic term (CSR-Score²). In model (1), the coefficient of CSR-Score is positive (β =0.232) and significant at the one percent level. This result suggests that after controlling for other factors, a 0.1 increase in CSR-Score is associated with an increase in the audit fees of 2.32 percent. Using the mean value of audit fees, which is \$4.3 million, a 2.32 percent premium translates into an increase of audit fees of around \$0.10 million for an average firm. However, when we add the quadratic term of CSR-Score in model (2), we can observe how the relationship between CSR performance and audit fees is first negative (for CSR-scores below 0.36) and then becomes positive as the CSR-Score becomes larger than 0.36. The coefficients of CSR-Score and CSR-Score squared are significant at the 1% level. This result is consistent with our arguments on the existence of an optimal level of CSR performance that

minimizes the auditor's assessment of the risk of material misstatement, such that an increase of a firm's CSR performance beyond (below) the optimal level increases (decreases) the auditor's assessment of the risk of material misstatement and as a result the audit fees.

[Table 3 about here]

About one third of the firms in our sample have a CSR-Score below 0.36 (level of CSR performance associated with minimal audit fees). Our findings are also economically relevant. For instance, an increase in the CSR-Score from 0 to 0.36 for an average firm is associated with an audit fees reduction of 9.35 percent. In contrast, an increase from 0.36 to 0.72 is associated with an increase in the audit fees by 9.11 percent. The Simunic model explains almost 84 percent of the total variation in the audit fees, and the results on the control variables largely resemble those reported in prior studies. In model (2), firm size is positive and highly significant, which indicates that larger clients pay higher audit fees. In addition, clients pay more when they have more inventory and account receivables, report losses, or have an auditor who is an industry specialist. In contrast, clients pay lower fees when they have a higher current ratio.

To reduce concerns related to omitted variables, we also estimate an enhanced audit fee model (following Ghosh and Lustgarten (2006) and Hogan and Wilkins (2008)). Coefficient estimates for this enhanced model are presented in models (3) and (4) of table 3. When the CSR-Score is introduced without its squared term (model (3)), we continue to find a positive coefficient (β =0.225), which is significant at the one percent level. In model (4), we add the quadratic term of CSR-Score and we find significant negative and positive coefficients for the variables CSR-Score and CSR-Score squared, respectively. Both coefficients are significant at the one percent level. This result confirms a U-shaped relationship between CSR performance and audit fees. In particular, for the estimated coefficients in model (4) of the CSR-Score (β =-0.455) and CSR-Score squared (β =0.648), audit fees are at a minimum when the CSR-Score is equal to 0.35. Any marginal increase in the CSR-Score below (beyond) this level is associated with reductions (increases) in the audit fees. The overall quadratic relationship between the CSR-Score and audit fees is illustrated in Figure 2. Regarding the control variables, we find that audit fees increase when the auditor is an industry specialist, when the audit takes place during

the auditor's busy season or when the client has more total assets, more current assets (relative to total assets) and extraordinary items and reports under US GAAP. In contrast, audit fees are lower when there was an auditor change or when the firm's current ratio is high.

[Figure 2 about here]

5.2. Contingencies

We propose that environmental dynamism, ownership concentration and leverage are likely to influence the relationship between CSR performance and audit fees. In particular, we argue that the positive implications of CSR performance for the reduction of the auditor's assessment of the risk of material misstatement are likely to increase with the degree of environmental dynamism, while the negative implications of CSR performance for the reduction of the auditor's assessment of the risk of material misstatement are likely to reduce with the degree of ownership concentration and leverage. We therefore expect the level of CSR performance associated with minimal audit fees, to be larger in dynamic environments, and for firms with high degrees of ownership concentration and debt. To test these conjectures, in Table 4, we employ the enhanced audit fees model and interact the variables CSRscore and CSR-Score squared with environmental dynamism in model (1), ownership concentration in model (2) and leverage in model (3). In line with our expectations, we find in model (1) that the quadratic relationship between CSR-Score and audit fees is contingent on environmental dynamism. The coefficient on the linear interaction is significant with a negative sign, while the coefficient on the quadratic-by-linear interaction is significant with a positive sign. From the estimates, we observe that the CSR-Score associated with minimal audit fees for the mean value of environmental dynamism (0.0091) is 0.366. In comparison, the CSR-Score associated with minimal audit fees for the mean value of environmental dynamism plus one standard deviation (0.0094) is 0.426, which is substantially higher. To further illustrate how environmental dynamism moderates the CSR performance-audit fees relationship, we show in Figure 3 how the level of CSR-Score associated with minimum audit fees moves upwards, as environmental dynamism becomes larger. These results provide support to the idea that the positive implications of CSR performance for the reduction of the auditor's assessment of the risk of material misstatement are further enhanced in dynamic environments.

[Table 4 and Figure 3 about here]

Next, Models (2) and (3) in table 4 examine whether ownership concentration and leverage, respectively, moderate the CSR performance–audit fees relationship. Consistent with our arguments we find that the quadratic relationship between CSR-Score and audit fees is contingent on ownership concentration and leverage. In both models, the coefficient on the linear interaction is significant with a negative sign, while the coefficient on the quadratic-by-linear interaction is significant with a positive sign. From the estimates, we obtain that the CSR-Score associated with minimal audit fees for the mean value of ownership concentration (0.255) is 0.299, while it moves to 0.386 for the mean value of ownership concentration plus one standard deviation (0.4724). Similarly, the estimates indicate that the optimal level moves from 0.354 for the mean value of leverage (0.554) to 0.393 for the mean value of leverage plus one standard deviation (0.740). Figures 4 and 5 illustrate these moderating effects, showing how the CSR-Score associated with minimal audit fees moves as ownership concentration and leverage become larger. This is in line with the idea that the presence of blockholders and debtholder reduces the negative implications of CSR performance for the reduction of the auditor's assessment of the risk of material misstatement.

[Figure 4 and 5 about here]

6. Sensitivity analysis

6.1 Sample splits for size and US/non-US

As in prior studies, we control for firm size using the natural logarithm of total assets. However, given the correlation between CSR-Score and size (which in our sample is 0.30), controlling for size may not be sufficient. To make certain that our results are not driven by size, we estimate the enhanced audit fees model for separate subsets of firms based on size (similar to Ghosh and Tang, 2015). We divide firms into two subsamples, depending on whether their total sales are below or above the industry-year average.¹⁸ The results from models (1) and (2) in Table 5 indicate that the coefficient of CSR-Score is negative and significant at the one percent level while the quadratic term is positive and

¹⁸ Our results are similar when using alternative size divisions, such as total assets.

significant at the one percent level for both size subgroups. In particular, for the smaller size group of firms (model (1)) the coefficients estimates of CSR-Score (β =-0.397) and of CSR-Score squared (β =0.586) suggest that the audit fees are at a minimum when the CSR-Score is equal to 0.34. Similarly, the audit fees of firms in the larger size group are at a minimum when the CSR-Score is equal to 0.35. Considering the similar findings for both groups of firms, these additional analyses indicate that the U-shape relationship between CSR performance and audit fees is robust to differences in firm size.

[Table 5 about here]

To verify that our results are not be driven by the relatively large proportion of US firms in our sample,¹⁹ we split our sample between US and non-US firms and present the results of our enhanced audit fee model for both subsamples in models (3) and (4) in Table 5. Given the international nature of the audit profession and the similarities in the audit regulation across countries, we expect our results not to be specific to the US setting. We find that the relationship between CSR-Score and audit fees is quadratic for both US and non-US firms, adding external validity to our results. Interestingly, the audit fees of firms in the US sample are at a minimum when the CSR-Score is equal to 0.27, while it is close to 0.38 for the non-US sample. This difference may relate to a stronger shareholder-oriented (versus stakeholder-oriented) institutional setting in the US, and suggest, as proposed by Aguilera and Jackson (2010) and Hoepner et al. (2016), that country characteristics or country sustainability may play a moderating role on the effect of CSR performance.

6.2 Alternative specifications of CSR performance

We conduct two additional analyses to examine the robustness of our results with respect to our measure of CSR performance. First, we examine the separate dimensions of CSR performance, namely the social and environmental performance, and show that our results are consistent to using the social and environmental scores. Second, we test our results when using D-SOCIAL-KLD. This CSR

¹⁹ About 37 percent of the firms in our sample have their headquarters in the US.

performance measure is only available for US listed companies, therefore we test the robustness of our results to the use of this measure in our subsample of US firms.

Our main measure of CSR performances combines the social and environmental performance scores, similar to Ioannou and Serafeim (2012) and Lys et al. (2015). Since our theoretical arguments apply to both the social and environmental dimensions, we expect our results to hold when considering these dimensions separately, i.e., we expect that our results are not driven by a single dimension. Models (1) and (2) in Table 6 present our results for the enhanced audit fee model when the environmental-Score and the social-Score are considered, respectively. The results across both dimensions are very similar. In line with our previous findings, there is a level of environmental and social performance associated with minimal audit fees. In particular, according to our results the audit fees are at a minimum for a level of environmental (social) performance equal to 0.35 (0.31). The finding that the environmental performance score associated with minimal audit fees is higher than the social performance score is in line with the idea developed in Hoepner et al. (2016) of a stronger effect of the corporate environmental performance dimension on the reduction of firms' default risk, compared to the social performance dimension.

Model (3) in table 6 shows the results for the enhanced audit fee model when using the D-SOCIAL-KLD measure of CSR performance. Similar to our previous findings, the coefficient of the D-SOCIAL-KLD is negative and significant, while the coefficient of the squared term is positive and significant, revealing a quadratic relationship between D-SOCIAL-KLD and audit fees. According to our results, the audit fees are at a minimum for a level of D-SOCIAL-KLD equal to 1.1. Indicatively, we note that about 26 percent of the sample has a D-SOCIAL-KLD score below 1.1. This result is similar to the results obtained for the CSR-Score when focusing on the US sample only (where 25 percent of the sample was found below the optimal level). Overall, the results for the D-SOCIAL-KLD measure of CSR performance provide additional validity to the results obtained when using our CSR-Score. Both measures are highly correlated, and both measures show a quadratic relationship with audit fees with a similar proportion of firms above and below the optimal level.

[Table 6 about here]

6.3 CSR performance and pre-audit misstatement risk

To provide additional support to our theoretical argument that there is a level of CSR performance that minimizes the auditor's assessment of the risk of material misstatement and thus the audit fees, we estimate a proxy of the pre-audit misstatement risk and analyse its relationship with CSR performance. As in Lobo and Zhao (2013), we rely on the misstatement detection model of Dechow et al. (2011) to estimate the pre-audit misstatement risk based on current year financial information and use the predicted probability of misstatements from that model as a proxy for the pre-audit misstatement risk.²⁰ Specifically, we model the pre-audit misstatement risk by using the following logistic specification:

 $Rest_{it} = \alpha + \beta_1 CSR \cdot Score_{it} + \beta_2 CSR \cdot Score_{it}^2 + \beta_3 Total \cdot Accrual_{it} + \beta_4 Drec_{it} + \beta_5 Dinv_{it} + \beta_6 Soft$ $assets_{it} + \beta_7 Dcsale_{it} + \beta_8 Droa_{it} + \beta_9 Issuance_{it} + \beta_{10} Abret_{it} + \beta_{11} Abret_{it-1} + \varepsilon_{it}$ (3)

The dependent variable *Rest* equals 1 if the annual report for the current year is subsequently restated, and 0 otherwise. *CSR-Score* and its quadratic term (*CSR-Score*²) are our independent variables of interest. *Total-accruals* is measured as the change in noncash assets (noncash total assets minus total liabilities and preferred stocks) from year t-1 to year t scaled by average total assets. *Drec* is the change in accounts receivables over total assets. *Dinv* is the change in inventory over total assets. *Soft-assets* are the proportion of soft assets (Total assets - Property, Plants and equipment - Cash and Short.Term Investments) to total assets. *Dcsale* is the percentage change in cash sales (cash sales = sales – the change in inventory). *Droa* is the change in return on assets. *Issuance* is a dummy variable taking the value of 1 if the firm issued debt or equity securities during year t, and zero otherwise. *Abret* is the annual buy-and-hold stock return minus annual buy-and-hold value weighted NYSE-AMEX-NASD index return.

[Table 7 about here]

Table 7, Panel A presents the descriptive statistics of the variables used in the analysis of the misstatement detection model (equation 3), while panel B shows the estimation results. The coefficients

²⁰We rely on Audit Analytics to obtain the restatement data, which is restricted to US listed firms. Thus, we are able to estimate the pre-audit misstatement risk and to analyses its relationship with CSR performance only for our subsample of US firms.

of our control variables show the same signs as Dechow et al. (2011) and Lobo and Zhao (2013) and the explanatory power of the model is also similar to the ones reported in these studies. Importantly, the coefficients on CSR-Score and CSR-Score squared are significant at the one percent level with a positive and negative signs, respectively. This result suggests a quadratic relationship between CSR performance and the pre-audit misstatement risk. Initially, the estimated pre-audit misstatement risk goes down as CSR-Score becomes larger, until the CSR-Score =0.38. Past this point, larger levels of the CSR-Score are associated with increases in the estimated pre-audit misstatement risk. Thus, our results suggest that there is a level of CSR performance associated with minimal pre-audit misstatement risk. Since auditors asses this pre-audit misstatement risk when analysing the risk of material misstatement this result provides further empirical support to our theoretical argument on the existence of a level of CSR performance that minimize the auditor's assessment of the risk of material misstatement and hence the audit fees.

7. Conclusions

The purpose of this paper is to investigate whether and how firms' CSR Performance influences the auditors' assessment of the risk of material misstatement, by analysing their audit pricing decision. Using an international sample of listed firms over the 2003-2012 period, we provide robust evidence of the existence of an optimal level of CSR performance associated with minimal audit fees. This result provides support to the idea that there is an optimal level of CSR performance that minimizes the auditors' assessment of the risk of material misstatement which in turn lowers the need for greater auditor effort; that is why auditors charge firms significantly less when their CSR performance is at the optimal level. We also show that the optimal level of CSR performance becomes larger in dynamic environments and for firms with higher degrees of ownership concentration and leverage. These results are in line with the idea that environmental dynamism (ownership concentration and leverage) further enhances (reduces) the positive (negative) implications of CSR performance for the reduction of the auditor's assessment of the risk of material misstatement. Prior auditing research has noted that risk assessment models may be refined and that examining the impact of various factors on audit judgements may assist the development of audit support system and decision aids (Asare and Davidson, 1995; Dechow et al, 1996; Bell and Carcello, 2000). The findings from this study extend our knowledge of auditors' judgement processes and may be useful to refine the traditional models of audit fees as well as client acceptance and retention guidelines. In addition, our findings suggest that audit effort is one mechanism through which CSR performance may influence the (post-audit) financial reporting quality of firms and, as a result, that audit fees may be an important omitted variable of prior studies that examine the effect of CSR performance on (post-audit) financial reporting quality.

Our results may also foster future research. In particular, future research could further enhance our understanding of the audit process by examining the impact of CSR performance on the auditor's engagement strategies such as assigning more high-risk specialists or industry experts, applying more intensive testing, and/or performing additional review. It would also be interesting to examine the likelihood of fraudulent financial reporting for an audit client, conditional on the level of CSR performance and to analyse the relationship between CSR performance and (post-audit) financial reporting quality controlling for the effect that CSR performance has on the audit effort.

In addition, future research could also apply our theoretical framework of the existence of positives and negatives implications of CSR performance for the auditor's assessment of the risk of material misstatements to explore the existence of a non-linear relationship between CSR performance and firm value or access to finance. Finding a non-linear relationship may help to reconcile the mixed results obtained in these two lines of literature (e.g., Renneboog et al., 2008; El Ghoul et al., 2011; Goss and Roberts, 2011; Cheng et al., 2014; Gregory et al., 2016; Hoepner et al., 2016). In this sense, future work could extend the study of Gregory et al. (2016) who show that firms with high CSR Performance have a higher value which they attribute to greater earnings persistence, and of Hoepner et al. (2016), who find that country sustainability reduces the cost of bank loans but that the effect of firm-level sustainability is not significant, by allowing for non-linearity on the effect of CSR performance. Finally, future research could explore other contingency factors, at country, industry or firm level that may influence the relationship between CSR performance and the auditors' assessment of the risk of material misstatement. As in Hoepner et al. (2016), it would be interesting to examine whether country characteristics or country sustainability moderate this relationship.

Variable name	Definition
Dependent variable	
Audit fees	Natural log of the fees (in US\$) charged by the auditor for the financial
	statements audit work.
CSR measures	
CSR-Score	The average of the ASSET4 benchmarked Social and Environmental Performanc
	scores. It takes values between zero and one.
Social-Score	ASSET4 benchmarked Social Performance score that takes values between zero
	and one. This measure reflects a company's capacity to generate trust and lovalty
	with its workforce customers and society through its use of best management
	practices
Environmental-Score	ASSET4 benchmarked Environmental Performance score that takes values
Environmental Score	hetween zero and one. This measure cantures a company's impact on living and
	between zero and one. This measure captures a company's impact on inving and
	non-niving natural systems, including the air, land and water, as well as completes
	ecosystems.
D-SOCIAL-KLD	A measure of CSR performance developed by Carroll et al. (2016). This measure
	builds on and improves the Kinder, Lydenberg, Domini and Co. (KLD) measure
	by weighting "hard" CSR activities more than "easy" CSR activities. This
	measure is only available for a subsample of US listed firms.
Contingency factors	
Environmental Dynamism	A standardized index of industry dynamism that is obtained by regressing
-	industry sales over the five years preceding the giving year (including the giving
	vear) against time, and using the standard error of the regression coefficient
	related to a time dummy variable divided by the average value of the industry's
	sales
Ownership Concentration	The total ownership stake held by blockholders (i.e. shareholders with at least fiv
o whership concentration	nercent of shares) obtained from Thomson One
Lavaraga	The ratio of the sum of the long term and short term debt to total assets
Leverage	The fatio of the sum of the folg-term and short-term debt to total assets.
Auditing characteristics	
Auditor-change	A dummy that takes the value of one when the client firm hires a new auditor and
	zero otherwise.
Auditor-tenure	The number of years of the audit engagement.
Auditor-specialization	Auditor's market share of the client's industry, defined as the sum of the total
	assets of all clients in an industry divided by the total assets of all firms in that
	industry during the year.
Busy-season	A dummy that takes the value of one when client's fiscal year-end is December of
	January and zero otherwise (with the exception of Japan, where it is March and
	April).
Firm controls	
Size	Natural log of the total assets of the firm (measured in thousands US\$).
Return-on-Assets	Return on assets defined as net income divided by total assets.
Receivables-and	The ratio of the sum of receivables and inventories divided by total assets.
Inventory-to- Total-assets	
Loss-in-current vear	A dummy that takes the value of one if the firm reports a loss during the current
2000 in carrone your	vear and zero otherwise
Current_ratio	The ratio of current assets to current liabilities
Current assets to Total	The ratio of current assets to total assets
	The momentum of color anomaly $\frac{1}{2}$ $\frac{1}{2}$ $\frac{1}{2}$ $\frac{1}{2}$ $\frac{1}{2}$ $\frac{1}{2}$ $\frac{1}{2}$ $\frac{1}{2}$
Sales-growth	I ne percentage of sales growth defined as $(sales_t - sales_{t-1}) / sales_{t-1}$.
Extraordinary-items	A dummy that takes the value of one if the firm reports discontinued operations
	and extraordinary items during the current year and zero otherwise.
Market-to-book-value	The ratio of the sum of the market value of common equity, the book values of
	preferred stock, and the book value of total debt to the book value of total assets.
H-index	Herfindahl index of industry sales concentration. It is the sum, per industry year,
	of the squared ratio of firm sales to total industry sales.
	A dummy that takes the value of one if the firm follows IERS and zero otherwise
IFRS	A duminy that takes the value of one if the firm follows if KS and zero otherwise
IFRS US-GAAP	A dummy that takes the value of one if the firm follows US GAAP and zero

Appendix A: Variable definitions

Appendix B: Description of the CSR performance scores from the Thomson Reuters Asset 4 Database.

Thomson Reuters ASSET4 covers more than 4.300 firms by assessing all firms listed on ASX 300, Bovespa, CAC 40, DAX, FTSE 250, MSCI Emerging Markets, MSCI World, NASDAQ 100, S&P 500, SMI and STOXX 600. Firms are assessed based on objective and publicly available data, which include stock exchange filings, CSR and annual reports, nongovernmental organizations' websites, and news sites. According to ASSET4, every data point goes through a multi-step verification and process control, which includes a series of data entry checks, automated quality rules and historical comparisons to ensure a high level of accuracy, timeliness and quality.

To build the performance scores for the different pillars, KPIs are obtained from over 750 individual data points, which are categorized as "drivers" or "outcomes. While drivers track policies that cover issues such as emission reduction or human rights, outcomes track quantitative results such as greenhouse gas emissions or personnel turnover. ASSET4 classifies the KPIs into categories within each pillar. For example, the environmental pillar consists of three category groupings: emission reduction, product innovation, and resource reduction. Each KPI is scored against the company peers between zero and one. To determine the relative weight of each KPI within each category, each KPI is assigned a Relative Level of Importance (RLI) from 0 to 5 based on several factors, such as the relevance of the KPI in the industry, whether it is derived from independent information content or the objective measurability of the KPI. The obtained weighted average scores for each category are then normalized and adjusted for skewness and the differential between the mean and the median, then fitted to a bell curve to derive ratings for each category between 0 and 1 for each company. The resulting category ratings are comparable across categories. Finally, the performance score of each pillar is the average of the different category ratings that make up the pillar, assuming equal weights for each category within the pillar.

In what follows we provide a detailed description of the different categories within the social and environmental pillars, as outlined in the ASSET 4 documentation.

Environmental Performa	nce Pillar
Resource Reduction	The resource reduction category measures a company's management commitment and
	effectiveness towards achieving an efficient use of natural resources in the production process. It
	reflects a company's capacity to reduce the use of materials, energy or water, and to find more
_	eco-efficient solutions by improving supply chain management.
Emission Reduction	The emission reduction category measures a company's management commitment and
	effectiveness towards reducing environmental emission in the production and operational
	processes. It reflects a company's capacity to reduce air emissions (greenhouse gases, F-gases,
	ozone-depleting substances, NOx and SOx, etc.), waste, hazardous waste, water discharges, spills
	or its impacts on biodiversity and to partner with environmental organizations to reduce the
	environmental impact of the company in the local or broader community.
Product Innovation	The product innovation category measures a company's management commitment and
	effectiveness towards supporting the research and development of eco-efficient products or
	services. It reflects a company's capacity to reduce the environmental costs and burdens for its
	customers, and thereby creating new market opportunities through new environmental
Social Daufaumanaa Dilla	technologies and processes or eco-designed, dematerialized products with extended durability.
Social Performance Pilla	
Employment Quanty	affectiveness towards providing high quality employment benefits and ich conditions. It reflects
	a company's capacity to increase its workforce loyalty and productivity by distributing rewarding
	and fair employment benefits, and by focusing on long-term employment growth and stability by
	promoting from within avoiding lay-offs and maintaining relations with trade unions
Health and Safety	The health and safety category measures a company's management commitment and
ficulti and Sulety	effectiveness towards providing a healthy and safe workplace. It reflects a company's capacity to
	increase its workforce lovalty and productivity by integrating into its day-to-day operations a
	concern for the physical and mental health, well-being and stress level of all employees.
Training and	The training and development category measures a company's management commitment and
Development	effectiveness towards providing training and development (education) for its workforce. It
	reflects a company's capacity to increase its intellectual capital, workforce loyalty and
	productivity by developing the workforce's skills, competences, employability and careers in an
	entrepreneurial environment.
Diversity and	The diversity and opportunity category measures a company's management commitment and
Opportunity	effectiveness towards maintaining diversity and equal opportunities in its workforce. It reflects a
	company's capacity to increase its workforce loyalty and productivity by promoting an effective
	life-work balance, a family friendly environment and equal opportunities regardless of gender,
	age, ethnicity, religion or sexual orientation.
Human Rights	The human rights category measures a company's management commitment and effectiveness
	towards respecting the fundamental human rights conventions. It reflects a company's capacity
	to maintain its license to operate by guaranteeing the freedom of association and excluding child,
	forced or compulsory labor.
Community	The community category measures a company's management commitment and effectiveness
	towards maintaining the company's reputation within the general community (local, national and
	global). It reflects a company's capacity to maintain its license to operate by being a good citizen
	(donations of cash, goods or staff time, etc.), protecting public health (avoidance of industrial
Customer / Droduct	The sustainer / product responsibility estadory measures a company's management as we it was the
Responsibility	and effectiveness towards creating value added products and services unholding the sustaments
Кезронзюниу	and encourseness towards creating value-added products and services upnotding the customer's security. It reflects a company's capacity to maintain its license to operate by producing quality.
	goods and services integrating the customer's health and safety and preserving its integrity and
	privacy also through accurate product information and labeling
	privacy also through accurate product information and facening.

References

- Aguilera, R. V., Jackson, G. 2010. Comparative and international corporate governance. The *Academy* of *Management Annals*, 4(1): 485–556.
- Asare, S., Davidson, R. 1995. Expectation of errors in unaudited book values: The effect of control procedures and financial condition. *Auditing: A Journal of Practice and Theory* 14: 1-18
- Atkinson, L., Galaskiewicz, J. 1988. Stock Ownership and Company Contributions to Charity, *Administrative Science Quarterly*, 33(1): 82-100.
- Barnea, A., Rubin, A. 2010. Corporate Social Responsibility as a Conflict between Shareholders. *Journal of Business Ethics* 97(1): 71–86.
- Barnett, M. L. 2007. Stakeholder influence capacity and the variability of financial returns to corporate social responsibility, *Academy of Management Review*, 32(3): 794-816.
- Barnett, M. L., Salomon, R. M. 2006. Beyond dichotomy: The curvilinear relationship between social responsibility and financial performance. *Strategic Management Journal* 27: 1101–1122.
- Baron, D. 2006. Businesses and its Environment, 5th Ed. Upper Saddle River, NJ. Prentice Hall.
- Baron, D. 2008. Managerial contracting and corporate social responsibility. *Journal of Public Economics* 92(1-2): 268–288.
- Bedard, J. C., Biggs, S. 1991. Pattern recognition, hypothesis generation, and auditor performance of analytical procedures. *The Accounting Review* 66: 635-55.
- Bedard, J., Johnstone, K. 2004. Earnings manipulation risk, corporate governance risk, and auditors' planning decisions. *The Accounting Review*, 79(2), 277–304.
- Behn, B., Choi, J. H., Kang, T. 2008. Audit quality and properties of analyst earnings forecasts. *The Accounting Review* 83(2): 327–359.
- Bell, T. B., Landsman, W.R., Shackelford, D.A., 2001. Auditors' perceived business risk and audit fees: Analysis and evidence. *Journal of Accounting Research* 39(1): 35–43.
- Bell, T. B., Carcello, J. 2000. A Decision Aid for Assessing the Likelihood of Fraudulent Financial Reporting. *Auditing: A Journal of Practice & Theory*, 19(1): 169-184.
- Berle, A., Means, G. 1932. The Modern Corporation and Private Property (Macmillan, New York).
- Berman, S. L., Wicks, A. C., Kotha, S. K. Jones, T. M. 1999. Does stakeholder orientation matter? The relationship between stakeholder management models and firm financial performance. *Academy of Management Journal* 42: 488–506.
- Boyd, B. K. 1990. Corporate Linkages and Organizational Environment: A Test of the Resource Dependence Model. *Strategic Management Journal*, 11: 419-430.
- Brammer, S., Millington, A. 2004. The development of corporate charitable contributions in the UK: A stakeholder analysis. *Journal of Management Studies* 41: 1411–1434.
- Butterworth, S., Houghton, K.A. 1995. Auditor switching: The pricing of audit services. *Journal of Business Finance and Accounting* 22 (3): 323–44.
- Campbell, J. L. 2007. Why Would Corporations Behave in Socially Responsible Ways? An Institutional Theory of Corporate Social Responsibility. *Academy of Management Review*, 32; 946-967.
- Carroll, A. B. 1979. A three-dimensional conceptual model of corporate social performance. *Academy* of Management Review 4: 497–506.
- Carroll, R. J., Primo, D. M., Richter, B. K. 2016. Using Item Response Theory to Improve Measurement in Strategic Management Research: An Application to Corporate Social Responsibility. *Strategic Management Journal* 37(1): 66-85.
- Cheng, B. Ioannou, I., Serafeim, G. 2014. Corporate social responsibility and access to finance. *Strategic Management Journal*, 35: 1-23.
- Cheng, I.-H., Hong, H., Shue, K. 2013. Do managers do good with other people's money? Working Paper no. 19432, NBER.
- Chih, H.-L., Shen, C.-H., Kang, F.-C. 2007. Corporate social responsibility, investor protection, and earnings management: Some international evidence. *Journal of Business Ethics* 79(1), 179-198.
- Choi, J., Wang, H., 2009. Stakeholder relations and the persistence of corporate financial performance. *Strategic Management Journal*, 30: 895-907.
- Christie, A. A., Zimmerman, J. 1994. Efficient and opportunistic choice of accounting procedures: corporate control contests. *The Accounting Review* 69: 539-566.
- Clarkson, P., Li, Y., Richardson, G., 2004. The market valuation of environmental expenditures by pulp and paper companies. *The Accounting Review* 79, 329–353.

- Cohen, J., Hanno, D. 2000. Auditors' Consideration of Corporate Governance and Management Control Philosophy in Preplanning and Planning Judgments. *Auditing: A Journal of Practice and Theory*, 19: 133-46.
- Colbert, J. 1988. Inherent risk: An investigation of auditors' judgements. *Accounting, Organisations and Society* 13(2): 111–121.
- Craswell, A.T., Francis, J.R., Taylor, S.L. 1995. Auditor brand name reputations and industry specializations. *Journal of Accounting and Economics* 20(3): 297–322.
- De Angelo, L. 1988. Managerial competition, information costs and corporate governance: The use of accounting performance measures in proxy contest. *Journal of Accounting and Economics* 10(1): 3-36.
- Dechow, P., Sloan, R., Sweeney, A. 1996. Causes and Consequences of Earnings Manipulation: An Analysis of Firms Subject to Enforcement Actions by the SEC, *Contemporary Accounting Research* 13: 1-36.
- Dechow, P., Ge, W., Schrand, C. 2010. Understanding earnings quality: A review of the proxies, their determinants and their consequences, *Journal of Accounting and Economics* 50(2): 344-401.
- Dechow, P. W., Ge, W., Larson, C., Sloan R. 2011. Predicting material accounting misstatements. *Contemporary Accounting Research* 28 (1): 17-82.
- DeFond, M., Zhang, J. 2014. A review of archival auditing research. A review of archival auditing research. *Journal of Accounting and Economics* 58(2), 275–326.
- Dess, G. G., Beard, D. W. 1984. Dimensions of Organizational Task Environments. *Administrative Science Quarterly* 29: 52-73.
- Di Giuli, A., Kostovetsky, L. 2014. Are red or blue companies more likely to go green? Politics and corporate social responsibility, *Journal of Financial Economics* 111: 158-180.
- Diamond, D. 1991, Monitoring and reputation: the choice between bank loans and directly placed debt, *Journal of Political Economy* 99: 689–721.
- Dye, R. 1993. Auditing standards, legal liability, and auditor wealth. *Journal of Political Economy*. 101(5): 887-914.
- Eccles, R. G., Ioannou, I., Serafeim, G. 2014. The impact of corporate sustainability on organizational processes and performance. *Management Science* 60(11), 2835–2857.
- Edmans, A. 2011. Does the stock market fully value intangibles? Employee satisfaction and equity prices. *Journal of Financial Economics* 101(3): 621–640.
- El Ghoul, S., Guedhami, O., Kwok, C.C.Y., Mishra, D.R. 2011. Does corporate social responsibility affect the cost of capital? *Journal of Banking and Finance*, 35 (9): 2388-2406.
- European Commission (Commission of the European Communities). 2001. Promoting a European Framework for Corporate Social Responsibilities, COM (2001) 366 final, Brussels.
- Farrell, J. 2001. Monopoly slack and competitive rigor. E. Rasmusen, ed. *Readings in Games and Information*. Blackwell Publishing,Oxford, UK.
- Fernandez-Kranz, D., Santalo, J. 2010. When necessity becomes a virtue: the effect of product market competition on corporate social responsibility. *Journal of Economics & Management Strategy* 19 (2), 453-487
- Flammer, C. 2015. Does corporate social responsibility lead to superior financial performance? A regression discontinuity approach. *Management Science* 61(11): 2549–2568.
- Fombrun, C.J. Shanley, M. What's in a name? Reputation building and corporate strategy. *Academy of Management Journal*, 33 (2): 233-258.
- Freeman, R. E. 1984. *Strategic Management: A stakeholder perspective*. Boston, MA: Piman, United States.
- Freeman, R. E., Harrison, J. S., Wicks, A. C. 2007. *Managing for stakeholders: survival, reputation, and success.* New Haven, CT: Yale University Press, United States.
- Friedman, M. 1970. The social responsibility of business is to increase its profits. *New York Times Magazine* 13 September.
- Frooman, J. 1999. Stakeholder influence strategies. Academy of Management Review 24: 191-206.
- Galbraith, J. A. 1973. *Designing Complex Organizations*. Addison-Wesley Longman Publishing, Boston, MA.
- Gao, F., Lisic, L.L., Zhang, I. 2014. Commitment to Social Good and Insider Trading. *Journal of Accounting and Economics* 57(2-3): 149–175.

- Gilson, S. 1990. Bankruptcy, boards, banks, and block holders. *Journal of Financial Economics* 27: 355-387.
- Ghosh A., Lustgarten, S. 2006. Pricing of Initial Audit Engagements by Large and Small Audit Firms. *Contemporary Accounting Research* 23(2): 333–368.
- Ghosh A., Tang, C.Y. 2015. Assessing financial reporting quality of family firms: The auditors' perspective. *Journal of Accounting and Economics* 60(1): 95–116.
- Glazer, A., Kannianinen, V., Poutvaara, P. 2008. Firm's ethics, consumer boykotts and signalling. Working Paper No. 2323, CESIFO.
- Godfrey, P. C. 2005. The relationship between corporate philanthropy and shareholder wealth: A risk management perspective. *Academy of Management Review* 30:777–798
- Godfrey, P. C., Merrill, C. B., Hansen, J. M. 2009. The relationship between corporate social responsibility and shareholder value: An empirical test of the risk management hypothesis. *Strategic Management Journal* 30(4): 425-445.
- Goss, A., Roberts, G.S. 2011. The Impact of Corporate Social Responsibility on the Cost of Bank Loans, *Journal of Banking and Finance* 35(7): 1794–1810.
- Graves, S. B., Waddock, S. A. 1994. Institutional owners and corporate social performance. *Academy* of Management Journal 37: 1034–1046.
- Greening, D. W., Turban, D. B. 2000. Corporate social performance as a competitive advantage in attracting a quality workforce. *Business & Society*.39: 254–280.
- Gregory, A., Collier, P. 1996. Audit fees and auditor change: An investigation of the persistence of fee reduction by type of change. *Journal of Business Finance and Accounting* 23 (1): 13–28.
- Gregory, A., Whittaker, J. 2013. Exploring the Valuation of Corporate Social Responsibility A Comparison of Research Methods, *Journal of Business Ethics* 116(1): 1–20.
- Gregory, A., Whittaker, J., Yan, X. 2016. Corporate Social Performance, Competitive Advantage, Earnings Persistence and Firm Value, *Journal of Business Finance and Accounting* 43(1-2): 3–30.
- Grossman, S., Hart, O. 1980. Takeover Bids, the Free Rider Problem and the Theory of the Corporation, *Bell Journal of Economics* 11(1): 42-64.
- Gul, F.A., Tsui, J.S.L. 1997. A test of the free cash flow and debt monitoring hypotheses: evidence from audit pricing. *Journal of Accounting and Economics*, 24 (2): 219–237.
- Hackenbrack, K., Knechel, W. R. 1997. Resource allocation decisions in audit engagements. *Contemporary Accounting Research* 14(3): 481-499.
- Hammersley, J. S., Bamber, E. M., Carpenter, T. D. 2010. The influence of documentation specificity and fraud risk priming on auditor fraud judgments and evidence evaluation decisions. *The Accounting Review* 85 (2): 547-571.
- Hay D., Knechel, W.R., Wong. N. 2006. Audit fees: A meta-analysis of the effect of supply and demand attributes. *Contemporary Accounting Research* 23 (1): 141–191.
- Hemingway, C.A., Maclagan, P.W. 2004. Managers' personal values as drivers of corporate social responsibility. *Journal of Business Ethics* 50(1): 33–44.
- Hillegeist, S. A. 1999. Financial Reporting and Auditing Under Alternative Damage Appointment Rules. *The Accounting Review* 74(3): 347-369.
- Hillman A. J., Keim, G. D. 2001. Shareholder value, stakeholder management, and social issues: what's the bottom line? *Strategic Management Journal* 22(2): 125-139.
- Hoepner, A., Oikonomou, I., Scholtens, B, Schröder, M. 2016. The Effects of Corporate and Country Sustainability Characteristics on The Cost of Debt: An International Investigation, *Journal of Business Finance and Accounting* 43(1-2): 158–190.
- Hogan, C.E., Wilkins, M.S. 2008. Evidence on the audit risk model: do auditors increase audit fees in the presence of internal control deficiencies? *Contemporary Accounting Research* 25(1), 219–242.
- Houston, R.W., Peters, M.F., Pratt, J.H., 1999. The audit risk model, business risk and audit planning decisions. *The Accounting Review*, 74(3): 281–298.
- Hribar, P., Kravet, T., Wilson, R. 2014. A new measure of accounting quality. *Review of Accounting Studies* 19 (1): 506–538.
- Huang, X.B., Watson, L. 2015. Corporate social responsibility research in accounting. *Journal of Accounting Literature*, 34: 1–16.
- Ioannou, I., Serafeim, G. 2012. What drives corporate social performance? The role of nation-level institutions. *Journal of International Business Studies* 34(9): 834–864.

- Ioannou, I., Serafeim, G. 2015. The impact of corporate social responsibility on investment recommendations. *Strategic Management Journal* 36(7): 1053–1081.
- Jansen, J. J. P., Van Den Bosch, F. A. J., Volberda, H. W. 2006. Exploratory innovation, exploitative innovation, and performance: effects of organizational antecedents and environmental moderators. *Management Science* 52(11): 1661 – 1674.
- Jensen, M., Meckling, W. 1976. Theory of the Firm: Managerial Behavior, Agency Costs and Ownership Structure. *Journal of Financial Economics* 3: 305-360.
- Jensen, M. C. 1986. Agency Costs of Free Cash-flow, Corporate finance and Takeovers, *American Economic Review* 76, 323-329.
- Jensen, M. C. 2001. Value maximization, stakeholder theory, and the corporate objective function. *Journal of Applied Corporate Finance*, 14(3): 8–21.
- Johnstone, K. M. 2000. Client acceptance decisions: Simultaneous effects of client business risk, audit risk, auditor business risk, and risk adaptation, *Auditing: A Journal of Practice and Theory* 19: 1–25.
- Johnstone, K. M., Bedard, J. C. 2001. Engagement planning, bid pricing, and client response in the market for initial attest engagements. *The Accounting Review*, 76(4): 199-220
- Johnstone, K. M., Bedard, J. C. 2003. Risk Management in client acceptance decisions. *The Accounting Review* 78(4): 1003-1025.
- Johnstone, K., Grambling, A., Rittenberg, L.E. 2014. Auditing: A Risk Based-Approach to Conducting a Quality Audit, 10th Edition, Cengage Learning.
- Jones R, Murrell A. 2001. Signaling positive corporate social performance: an event study of familyfriendly firms. *Business and Society* 40(1): 59–78.
- Ke, B., Petroni, K., Safieddine, A. 1999. Ownership concentration and sensitivity of executive pay to accounting performance measures: Evidence from publicly and privately-held insurance companies, *Journal of Accounting and Economics* 28, 185-209.
- Kim, Y., Park, M. S., Wier, B. 2012. Is Earnings quality associated with corporate social responsibility? *The Accounting Review* 87(3): 761-796.
- Kizirian, T.G., Mayhew, B.W., Sneathen, L.D. 2005. The impact of management integrity on audit planning and evidence. *Auditing: A Journal of Practice and Theory* 24(2): 49–67.
- Levinthal, D., Myatt, J. 1994. Co-Evolution of Capabilities and Industry: The Evolution of Mutual Fund Processing, *Strategic Management Journal* 15: 45-62.
- Lobo, G., Zhao, Y. 2013. Relation between Audit Effort and Financial Report Misstatements: Evidence from Quarterly and Annual Restatements. *The Accounting Review* 88 (4): 1385-1412.
- Luo, X., Wang, H., Raithel, S., Zheng, Q. 2015. Corporate social performance, analyst stock recommendations, and firm future returns. *Strategic Management Journal* 36(1), 123-136
- Lyon, J.D., Maher, M.W. 2005. The importance of business risk in setting audit fees: Evidence from cases of client misconduct. *Journal of Accounting Research* 43(1), 133–151.
- Lys, T., Naughton, J., Wang, C. 2015. Signaling through corporate accountability reporting. *Journal of Accounting and Economics* 60(1): 56–72.
- Mackey, A., Mackey, T., Barney, J. 2007. Corporate social responsibility and firm performance: Investor preferences and corporate strategies. *Academy of Management Review* 32(3): 817–835.
- Margolis, J., Elfenbein, H., Walsh, J. 2007. Does it pay to be good? A meta-analysis and redirection of research on the relation between corporate social and financial performance. Working Paper. Harvard University.
- Matsumura, E. M., Tucker, R. 1992. Fraud Detection: A Theoretical Foundation. *The Accounting Review* 67(4): 753-782.
- Maxwell, J. W., Lyon, T. P., Hackett, S. C. 2000. Self-Regulation and Social Welfare: The Political Economy of Corporate Environmentalism. *Journal of Law and Economics*, 43(2): 583-618.
- McConnell, J. J., Servaes, H. 1990. Additional evidence on equity ownership and corporate value. *Journal of Financial Economics* 27, 595-612.
- McWilliams, A., Siegel, D., Wright, P.M. 2006. Corporate Social responsibility: Strategic implications. *Journal of Management Studies*, 43(1): 1–18.
- McWilliams, A., Siegel, D. 2001. Corporate social responsibility: A theory of the firm perspective. *Academy of Management Review*, 26(1), 117–127.

- Milliken, F. J. 1987. Three types of perceived uncertainty about the environment: State, effect, and response uncertainty. *Academy of Management Review*, 12, 133–143.
- Minor, D. B., Morgan, J. 2011. CSR as Reputation Insurance: Primum Non Nocere, *California Management Review* 53(3): 40–59.
- Nielsen. 2014. Doing well by doing good.
- http://www.nielsen.com/content/dam/nielsenglobal/jp/docs/report/2014/Nielsen%20Global%20Corporate%20Social%20Responsibility%20Report%20-%20June%202014.pdf
- Nielsen. 2015. Global Corporate Sustainability Report.
- https://www.nielsen.com/content/dam/nielsenglobal/dk/docs/global-sustainability-report-oct-2015.pdf
- O'Keefe, T. B., Simunic, D. A., Stein, M. T. 1994. The production of audit services: Evidence from a major public accounting firm. *Journal of Accounting Research* 32(2): 241–261.
- Orlitzky, M., Schmidt, F.L., Rynes, S.L. 2003. Corporate social and financial performance: A metaanalysis. *Organization Studies*, 24(3): 403–441.
- Peloza, J. 2009. The challenge of measuring financial impacts from investments in corporate social performance. *Journal of Management* 35: 1518–1541.
- Petrovits, C. M. 2006. Corporate-sponsored foundations and earnings management. *Journal of* Accounting and Economics, 41(3): 335-362.
- Pfeffer, J., Salancik, G. R. 1978. *The external control of organizations: a resource dependence perspective*. New York: Harper and How.
- Plumlee, M., Brown, D., Marshall, S. 2015. The impact of voluntary environmental disclosure quality on firm value. *Journal of Accounting and Public Policy* 34(4): 336-361.
- Porter, M.E., Kramer, M.P. 2006. Strategy and Society: The Link Between Competitive Advantage and Corporate Social Responsibility, *Harvard Business Review*, December: 1–13.
- Pratt, J., Stice, J. 1994. The effect of client characteristics on auditor litigation risk judgments, required audit evidence, and recommended audit fees. *The Accounting Review* 69: 639-656.
- Prior, D., Surroca, J., Tribó, J.A. 2008. Are socially responsible managers really ethical? Exploring the relationship between earnings management and corporate social responsibility. *Corporate Governance: An international Review* 16(3): 160–177.
- Renneboog, L., Ter Horst, J., Zhan, C. 2008. Socially Responsible Investments: Institutional Aspects, Performance, and Investor Behaviour, *Journal of Banking & Finance* 32(9): 1723–42.
- Rice, S. C., and D. P. Weber. 2012. How Effective Is Internal Control Reporting under SOX 404? Determinants of the (Non-) Disclosure of Existing Material Weaknesses. *Journal of Accounting Research* 50 (3): 811-843.
- Rubin, P.H. 1990. Managing Business Transactions. The Free Press, Maxwell Macmilian International, New York.
- Schmitz, J., Schrader, J. 2015. Corporate social responsibility: A microeconomic review of the literature. *Journal of Economic Surveys*, 29(1), 27-45.
- Shelton, S., Whittington, O., Landsittel, D. 2001. Auditing firms' fraud risk assessment practices. *Accounting Horizons* 15(1), 19–33.
- Shibano, T. 1990. Assessing audit risk from errors and irregularities. *Journal of Accounting Research*, 28: 110–140.
- Shiu, Y.-M., Yang, S.-L. 2017. Does engagement in corporate social responsibility provide strategic insurance-like effects? Strategic Management Journal 38(2): 455–470.
- Shleifer, A., Vishny, R. 1997. A survey of corporate governance. Journal of Finance 52: 737-783.
- Simunic, D., 1980. The pricing of audit services: Theory and evidence. *Journal of Accounting Research* 18(1): 161–190.
- Sirmon, D. G., Hitt, M. A., Ireland, R. D. 2007. Managing firm resources in dynamic environments to create value: looking inside the black box. *Academy of Management Review* 32(1): 273–292.
- Solomon, I., Shields, M., Whittington, O. 1999. What do industry-specialist auditors know? *Journal of Accounting Research* 30: 191-208.
- Srinidhi, B. N., Gul, F. A. 2007. The Differential Effects of Auditors' Nonaudit and Audit Fees on Accrual Quality. *Contemporary Accounting Research* 24 (2):595-629.
- Stulz, R. 1990. Managerial Discretion and Optimal Financing Policies. *Journal of Financial Economics* 26: 3-27

- Surroca, J., Tribo, J.A. 2008. Managerial entrenchment and corporate social performance. *Journal of Business Finance and Accounting* 35, 748–789.
- Vanhamme, J., Grobben, B. 2009. "Too good to be true!" The effectiveness of CSR history in countering negative publicity. *Journal of Business Ethics*, 85, 273-283
- Waddock, S. 2002. *Leading Corporate Citizens: Vision, Values, Value Added*. New York: McGraw-Hill.
- Waddock, S. 2008. Building a new institutional infrastructure for corporate responsibility. Academy of Management Perspectives, 22(3): 87–108.
- Waddock, S., Graves, S. B. 1997. The corporate social performance financial performance link. *Strategic Management Journal* 18(4): 303–319.
- Wang, H., Choi, J., Li, J. 2008. Too little or too much? Untangling the relationship between corporate philanthropy and firm financial performance. *Organization Science* 19: 142-159.
- Warfield, T. D., Wild, J. J., Wild, K. L. 1995. Managerial ownership, accounting choices, and informativeness of earnings. *Journal of Accounting and Economics* 20 (1): 61 91.
- Wilks, T., Zimbelman, M. 2004. Decomposition of Fraud-risk Assessments and Auditors' Sensitivity to Fraud Cues. *Contemporary Accounting Research* 21 (3): 719-745.
- Wright, P, Ferris S. 1997. Agency conflict and corporate strategy: The effect of divestment on corporate value. *Strategic Management Journal* 18(1): 77–83.



Figure 1. The Hypothesized U-shaped Relationship between CSR performance and Audit Fees

This figure plots the level of audit fees for different values of the CSR performance. Curve A shows how the positive implications of CSR performance for the reduction of the auditor's assessment of the risk of material misstatement influences the firm's audit fees for different levels of CSR performance. Curves B, C and D show how the negative implications of CSR performance for the reduction of the auditor's assessment of the risk of material misstatement, in terms of direct cost, agency concerns and concerns about the firm's risk of failing to deliver upon the CSR promises, respectively, influence the firm's audit fees for different levels of CSR performance. Curve E reflect the net effects of the firm's CSR performance on the audit fees for different levels of CSR performance.

Figure 2. The relationship between CSR performance and Audit Fees



This figure plots, for different values of the CSR-Score, the results from specification (4) in Table 3, which estimates the enhanced audit fee model. Specifically, it plots the function: y = 15.24 - 0.455*CSR-Score + 0.648*CSR-Score². The intercept (15.24) is computed by using the estimated coefficients from specification (4) in Table 3 and the mean values of the rest of variables included (see Table 1). By taking the derivative of the function, we find the minimum at CSR-Score = 0.35.



Figure 3: CSR Performance- Audit Fees: Contingent on Environmental Dynamism

This figure plots, for different values of the CSR-Score, the results from specification (1) in Table 4. Specifically, it plots the function: y = 15.19 + 7.437*CSR-Score - 5.705*CSR-Score² + 131* Environmental Dynamism - 873* CSR-Score* Environmental Dynamism +703* CSR-Score²* Environmental Dynamism The intercept (15.19) is computed by using the estimated coefficients from specification (1) in Table 4 and the mean values of the rest of variables included (see Table 1).



Figure 4: CSR Performance- Audit fees: Contingent on Ownership Concentration

This figure plots, for different values of the CSR-Score, the results from specification (2) in Table 4. Specifically, it plots the function: y = 15.22 - 0*CSR-Score + 0.302*CSR-Score² + 0*Ownership Concentration - 1.447* CSR-Score* Ownership Concentration + 1.235* CSR-Score²* Ownership Concentration

The intercept (15.22) is computed by using the estimated coefficients from specification (2) in Table 4 and the mean values of the rest of variables included (see Table 1).



Figure 5: CSR Performance - Audit fees: Contingent on Leverage

This figure plots, for different values of the CSR-Score, the results from specification (3) in Table 4. Specifically, it plots the function: y = 15.18 + 0.664*CSR-Score - 0.594*CSR-Score² + 0.285*Leverage - 2.006* CSR-Score * Leverage +2.213* CSR-Score²* Leverage

The intercept (15.18) is computed by using the estimated coefficients from specification (2) in Table 4 and the mean values of the rest of variables included (see Table 1).

1 auto	e 1. Descript	ive statistics	•		
Variable	Obs.	Mean	Std. Dev.	Min	Max
Audit fees	12330	15.2386	1.8275	0.5878	19.9976
CSR-Score	12330	0.5657	0.2871	0.0597	0.9781
Social-Score	12330	0.5666	0.2965	0.0341	0.9888
Environmental-Score	12330	0.5648	0.3140	0.0831	0.9719
D-SOCIAL-KLD	4474	3.8443	2.6376	-4.8082	10.4796
Environmental Dynamism	12330	0.0091	0.0003	0.0053	0.0094
Ownership Concentration	12330	0.2548	0.2176	0.0000	0.9021
Leverage	12330	0.5540	0.1863	0.0231	1.1188
Auditor-Change	12330	0.0380	0.1911	0.0000	1.0000
Auditor-Tenure	12330	8.6753	3.8425	1.0000	15.0000
Auditor-Specialization	12330	0.2528	0.1921	0.0001	1.0000
Busy-Season	12330	0.7319	0.4430	0.0000	1.0000
Size	12330	16.0993	2.1579	8.6545	24.9621
Return-on-Assets	12330	0.0560	0.0709	-0.4960	0.3198
Receivables-and Inventory-to- Total-assets	12330	0.2673	0.1767	0.0000	0.9926
Loss-in-Current-Year	12330	0.1041	0.3054	0.0000	1.0000
Current-Ratio	12330	1.7472	1.2265	0.3000	14.2900
Current-assets-to-Total-assets	12330	0.4392	0.2357	0.0147	1.4723
Sales-Growth	12330	0.0865	0.2075	-0.5284	2.9963
Extraordinary-items	12330	0.1131	0.3168	0.0000	1.0000
Market-to-Book-Value	12330	2.9172	3.7953	-49.0715	49.2226
H-Index	12330	0.1932	0.1436	0.0500	1.0000
IFRS	12330	0.3537	0.4781	0.0000	1.0000
US-GAAP	12330	0.3998	0.4899	0.0000	1.0000

Table 1 Decorrintive statistics

This table describes the main variables used for the analysis. All variables span the 2003-2012 period and correspond to 12,330 firm-year observations from 28 countries. Variable definitions: Audit fees is the natural log of the fees paid to auditor for the financial statement audit work (in US\$). CSR-Score is the average of the ASSET4 benchmarked Social and environmental Performance scores. It takes values between zero and one. Social-Score is the ASSET4 benchmarked Social performance score that takes values between zero and one. It reflects a company's capacity to generate trust and loyalty with its workforce, customers and society, through its use of best management practices. Environmental Score is the ASSET4 benchmarked Environmental performance score that takes values between zero and one. It captures a company's impact on living and nonliving natural systems, including the air, land and water, as well as complete ecosystems. D-SOCIAL-KLD is the CSR performance measure developed by Carroll et al. (2016) that builds on and improves the Kinder, Lydenberg, Domini and Co. (KLD) measure; Environmental dynamism is a standardized index of industry dynamism obtained by regressing industry sales over the five years preceding the giving year (including the giving year) against time, and using the standard error of the regression coefficient related to a time dummy variable divided by the average value of the industry's sales. Ownership concentration is the total ownership stake held by blockholders. Leverage is the ratio of the sum of the long-term and shortterm debt to total assets. Auditor-change is a dummy that takes the value of one when the client contracts a new auditor and zero otherwise. Auditor-tenure reflects the number of years of the audit engagement. Auditor-specialization is the auditor's market share of the client's industry, defined as the sum of the total assets of all clients in an industry divided by the total assets of all firms in that industry during the year. Busy-season is a dummy that takes the value of one when client's fiscal year-end is December or January and zero otherwise (with the exception of Japan, where it is March and April); Size: natural log of the total assets of the firm (in thousands US\$). Leverage is the ratio of the sum of the long-term and short-term debt to total assets. Return-on-Assets is defined as net income divided by total assets. Receivables-and-Inventory-to-Total-assets is the ratio of the sum of receivables and inventories divided by total assets. Loss-in-current-year is a dummy that takes the value of one if the firm reports a loss during the current year, and zero otherwise. Current-ratio is the ratio of current assets to current liabilities. Current-assets-to-Total-assets is the ratio of current assets to total assets; Sales-growth is the percentage of sales growth defined as: (salest - salest-1) / salest-1. Extraordinary-items is a dummy that takes the value of one if the firm reports discontinued operations and extraordinary items during the current year and zero otherwise. Market-to-book-value is the ratio of the sum of the market value of common equity, the book values of preferred stock, and the book value of total debt to the book value of total assets. H-index is the Herfindahl index of industry sales concentration. It is the sum at industry year of the squared ratio of firm sales to total industry sales. IFRS is a dummy that takes the value of one if the firm follows IFRS and zero otherwise. US-GAAP is a dummy that takes the value of one if the firm follows US GAAP and zero otherwise.

		1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16
1	Audit fees	1.00															
2	CSR-Score	0.25	1.00														
3	Social-Score	0.16	0.94	1.00													
4	Environmental-Score	0.31	0.94	0.77	1.00												
5	D-SOCIAL-KLD	0.60	0.73	0.69	0.70	1.00											
7	Environmental Dynamism	0.09	0.07	0.06	0.07	0.06	1.00										
7	Ownership Concentration	-0.19	-0.09	-0.05	-0.12	-0.16	-0.16	1.00									
8	Leverage	0.04	0.19	0.18	0.18	0.25	0.01	-0.02	1.00								
9	Size	0.87	0.30	0.20	0.35	0.69	0.09	-0.15	0.09	1.00							
10	H-Index	-0.03	-0.06	-0.05	-0.06	0.00	-0.07	0.01	-0.12	0.00	1.00						
11	Current-Ratio	-0.06	-0.18	-0.19	-0.16	-0.24	0.02	-0.02	-0.56	-0.13	0.09	1.00					
12	Receivables-and Inventory-to-Total-Assets	0.05	0.08	0.06	0.10	-0.05	-0.08	0.00	0.07	-0.07	-0.17	0.07	1.00				
13	Return-on-Assets	-0.11	-0.01	0.03	-0.04	0.03	-0.03	0.02	-0.26	-0.14	-0.01	0.11	0.12	1.00			
14	Loss-in-Current-Year	-0.02	-0.06	-0.07	-0.04	-0.07	0.00	0.01	0.08	-0.03	0.06	0.03	-0.07	-0.61	1.00		
15	Auditor-Tenure	0.02	0.04	0.03	0.05	0.09	0.18	-0.26	-0.03	-0.01	0.05	0.05	-0.03	0.03	-0.02	1.00	
16	Auditor-Specialization	0.02	0.00	0.00	0.01	0.07	-0.01	-0.03	-0.01	0.02	0.13	0.01	-0.01	-0.01	0.00	0.02	1.00

Table 2. Correlations

This table presents the pairwise correlations of the main variables used for the analyses. All variables are defined in Table 1 and Appendix A.

	Simunic model		Enhanced model		
	(1)	(2)	(3)	(4)	
CSR-Score	0.232***	-0.516***	0.225***	-0.455***	
	(0.033)	(0.128)	(0.032)	(0.125)	
CSR-Score ²	(0.055)	0 712***	(0.052)	0 648***	
		(0.119)		(0.117)	
Size	0 614***	0.607***	0.612***	0.606***	
Size	(0.014)	(0.007)	(0.012)	(0.000)	
H-Index	(0.000)	(0.000)	(0.000)	(0.000)	
11-Index	(0.210)	(0.210)	(0.219)	(0.219)	
Current Ratio	(0.219)	(0.219)	-0.068***	0.219	
Current-Ratio	-0.033	-0.037	-0.008	(0.007)	
Passivelas and Inventory to Total agests	(0.000) 0.754***	(0.000)	(0.007)	(0.007)	
Receivables-and inventory-to-rotar-assets	(0.052)	(0.052)	(0.091)	(0.086)	
Datum on Assats	(0.033)	(0.033)	(0.080)	(0.080)	
Neturii-Oii-Assets	(0.100)	(0.134)	(0.121)	(0.121)	
Lass in Comment Vacu	(0.122)	(0.122)	(0.131)	(0.131)	
Loss-in-Current- y ear	$0.0/1^{**}$	0.00/ ^{**}	$0.04/^{*}$	0.044	
	(0.028)	(0.028)	(0.028)	(0.028)	
Auditor-lenure	0.002	0.002	-0.001	-0.001	
	(0.002)	(0.002)	(0.002)	(0.002)	
Auditor-Specialization	0.100***	0.099***	0.118***	0.118***	
	(0.034)	(0.033)	(0.033)	(0.033)	
Leverage			0.03	0.033	
			(0.050)	(0.050)	
Current-Assets-to-Total-Assets			0.762***	0.755***	
			(0.067)	(0.067)	
Sales-Growth			-0.029	-0.03	
			(0.039)	(0.039)	
Market-to-Book-Value			0.002	0.002	
			(0.002)	(0.002)	
Extraordinary-Items			0.255***	0.255***	
			(0.020)	(0.020)	
Busy-Season			0.114***	0.112***	
			(0.017)	(0.016)	
Auditor-Change			-0.148***	-0.149***	
C			(0.043)	(0.043)	
IFRS			0.02	0.019	
			(0.040)	(0.040)	
US-GAAP			0.255***	0.255***	
			(0.070)	(0.071)	
Country dummies	ves	ves	ves	ves	
Industry dummies	ves	ves	ves	ves	
Year dummies	ves	ves	ves	ves	
Constant	4 801***	5 052***	4 750***	4 987***	
Constant	(0.272)	(0.275)	(0.281)	(0.284)	
R-squared	0.838	0.838	0.844	0.20 + j	
Observations	12330	12330	12330	12330	

 Table 3. CSR Performance and Audit Fees

Deservations12330123301233012330* p < 0.10, ** p < 0.05, *** p < 0.01. The dependent variable for all specifications is the natural logarithm of Auditfees. Specifications (1) and (2) report the regression results bases on the Simunic (1980) audit fee model.Specifications (3) and (4) report the regression results based on our enhanced model of audit fees, based on Hayet al. (2006) and Hogan and Wilkins (2008). All variables are defined in Table 1 and Appendix A. Standarderrors are reported in parentheses.

	c and Muult Pee	s – Contingency	lactors
	(1)	(2)	(3)
CSR-Score	7 437**	-0.059	0.664*
	(2,05())	(0.101)	(0.257)
	(3.056)	(0.191)	(0.357)
CSR-Score ²	-5.705**	0.302*	-0.594*
	(2.883)	(0.174)	(0.338)
Environmental Dynamism ¹	0.131*	(• • • • • • • • • • • • • • • • • • •	(
	0.131		
	(0.074)		
Environmental Dynamism [†] * CSR-Score	-0.873***		
	(0.337)		
\mathbf{F} : $\mathbf{H}\mathbf{P}$: $\mathbf{H}\mathbf{F}$ (CCP C)	(0.337)		
Environmental Dynamism ¹ * CSR-Score ²	0./03**		
	(0.318)		
Ownership Concentration		0 134	
Ownership Concentration		(0.124)	
		(0.124)	
Ownership Concentration * CSR-Score		-1.447***	
		(0.553)	
Ormanitin Concentration * CCD Com?		1 225**	
Ownership Concentration * CSR-Score ²		1.233***	
		(0.506)	
Leverage	0.030	0.030	0 285**
Levelage	(0, 050)	(0.050)	(0.140)
	(0.030)	(0.030)	(0.140)
Leverage * CSR-Score			-2.006***
			(0.631)
Laverage $*$ CSP Score ²			2 212***
Levelage CSR-Scole			2.213
			(0.588)
Size	0.605***	0.603***	0.606***
	(0, 007)	(0, 007)	(0, 007)
TT ' 1	(0.007)	(0.007)	(0.007)
H-index	-0.12	-0.18	-0.208
	(0.230)	(0.225)	(0.223)
Current Ratio	0.060***	-0.070***	-0.071***
Current-Katio	-0.009	-0.070	-0.071
	(0.007)	(0.007)	(0.007)
Receivables-and Inventory-to- Total-Assets	0.116*	0.111*	0.126*
·	(0.067)	(0.067)	(0.067)
Determ on Accest	0.007	0.021	0.002
Keturn-on-Assets	0.024	0.031	0.003
	(0.134)	(0.135)	(0.134)
Loss-in-Current-Year	0.044	0.041	0.043
	(0, 0, 29)	(0,029)	(0, 0, 29)
	(0.028)	(0.028)	(0.028)
Auditor-Tenure	-0.001	-0.002	-0.001
	(0,002)	(0.002)	(0.002)
Auditor Spacialization	0 10027	0 101***	0 110***
Auditor-Specialization	0.122	0.121	0.119
	(0.035)	(0.035)	(0.035)
Current-Assets-to-Total-Assets	0.752***	0.739***	0.751***
	(0, 053)	(0,053)	(0,053)
	(0.033)	(0.055)	(0.055)
Sales-Growth	-0.031	-0.029	-0.035
	(0.034)	(0.035)	(0.034)
Market-to-Book-Value	0.002	0.002	0.002
Market to Book Value	(0,002)	(0.002)	(0,002)
	(0.002)	(0.002)	(0.002)
Extraordinary-Items	0.253***	0.252***	0.256***
	(0.022)	(0.022)	(0.022)
Duay Saaaan	0.110***	0.100***	0.112***
Busy-Season	0.110***	0.109***	0.113
	(0.017)	(0.017)	(0.017)
Auditor-Change	-0 148***	-0 156***	-0 150***
Auditor Change	(0.028)	(0.029)	(0.028)
	(0.038)	(0.038)	(0.038)
IFRS	0.020	0.030	0.020
	(0.032)	(0.032)	(0.032)
US CAAD	0.255***	0 272***	0.262***
US-UAAF	0.233	0.2/3	0.203
	(0.053)	(0.054)	(0.053)
Country dummies	ves	ves	ves
Industry dummies	, Voc	, VAC	, Voc
	yus	yus	yus
Y ear dummies	yes	yes	yes
Constant	2.285**	4.868***	3.736***
	(1 0 1 1)	(0.932)	(0.809)
D	0.044	0.9321	0.0077
K-squared	0.844	0.845	0.844
Ν	12330	12330	12330

 Table 4. CSR Performance and Audit Fees – Contingency factors

* p < 0.10, ** p < 0.05, *** p < 0.01. The dependent variable for all specifications is the natural logarithm of Audit fees. Specifications (1), (2) and (3) report the regression results based on the enhanced model of audit fees considering the environmental dynamism, ownership concentration and leverage as contingency factors, respectively. All variables are defined in Table 1 and Appendix A. Standard errors are reported in parentheses. ¹ The coefficients shown are multiplied by 10^{-3} .

	05/11011-05			
	Smaller than	Larger than		
	median yearly	median yearly	US	Non-US
	industry sales	industry sales	sample	sample
	(1)	(2)	(3)	(4)
CSR-Score	-0.397**	-0.455***	-0.375**	-0.404**
	(0.186)	(0.164)	(0.146)	(0.185)
CSR-Score ²	0.586***	0.655***	0.702***	0.530***
	(0.177)	(0.149)	(0.134)	(0.171)
Size	0.607***	0.611***	0.537***	0.647***
	(0.012)	(0.010)	(0.009)	(0.011)
H-index	0.018	-0.436	-0.055	-0.192
	(0.310)	(0.388)	(0.314)	(0.283)
Current-Ratio	-0.053***	-0.087***	-0.075***	-0.064***
	(0.009)	(0.010)	(0.009)	(0.010)
Receivables-and Inventory-to-Total-Assets	0.147	0.124	0.436***	-0.128
	(0.124)	(0.119)	(0.127)	(0.120)
Return-on-Assets	0.069	-0.061	0.009	0.029
	(0.177)	(0.195)	(0.157)	(0.203)
Loss-in-Current-Year	0.087**	-0.004	0.100***	0.012
	(0.041)	(0.037)	(0.035)	(0.039)
Auditor-Tenure	0.000	-0.003	-0.012***	0.006*
	(0.004)	(0.003)	(0.003)	(0.003)
Auditor-Specialization	0.079*	0.168***	0.133***	0.064
	(0.046)	(0.048)	(0.038)	(0.050)
Leverage	0.110	-0.093	0.220***	-0.092
	(0.075)	(0.067)	(0.057)	(0.076)
Current-Assets-to-Total-Assets	0.718***	0.777***	0.591***	0.922***
	(0.093)	(0.092)	(0.096)	(0.095)
Sales-Growth	-0.068	0.003	-0.061	0.007
	(0.058)	(0.051)	(0.047)	(0.053)
Market-to-Book-Value	0.005*	-0.002	0.000	0.002
	(0.003)	(0.003)	(0.002)	(0.003)
Extraordinary-Items	0.243***	0.252***	0.236***	0.265***
	(0.028)	(0.027)	(0.022)	(0.033)
Busy-Season	0.092***	0.132***	0.065***	0.134***
	(0.023)	(0.023)	(0.021)	(0.024)
Auditor-Change	-0.106*	-0.199***	-0.172**	-0.112**
	(0.064)	(0.056)	(0.074)	(0.051)
IFRS	0.077	-0.161**		
	(0.048)	(0.072)		
US-GAAP	0.272***	0.166*		
	(0.101)	(0.092)		
Country dummies	yes	yes	no	yes
Industry dummies	yes	yes	yes	yes
Year dummies	yes	yes	yes	yes
Constant	5.515***	3.762***	5.524***	3.417***
	(0.414)	(0.527)	(0.246)	(0.398)
R-squared	0.797	0.880	0.670	0.867
Observations	6130	6200	4709	7621

Table 5. Robustness: CSR Performance and Audit Fees splitting the sample by size and US/non-US

 $\frac{6130}{6200} \frac{6200}{4709} \frac{7621}{7621}$ * p<0.10, ** p<0.05, *** p<0.01. The dependent variable for all specifications is the natural logarithm of Audit fees. Specification (1) and (2) report the regression results based on the enhanced model of audit fees for firms that have below median yearly industry sales (i.e., smaller firms in the industry) and above median yearly industry sales (i.e., larger firms in the industry), respectively. Specification (3) estimates the enhanced audit fee model only for US firms. Specification (4) estimates the enhanced audit fee model for non-US firms. All variables are defined in Table 1 and Appendix A. Standard errors are reported in parentheses.

Perfo	rmance		
	(1)	(2)	(3)
Environmental-Score	-0.391***	(=)	(0)
	(0.121)		
Environmental-Score ²	0.567***		
	(0.115)		
Social-Score		-0.254**	
		(0.112)	
Social-Score ²		0.408***	
		(0.105)	
D-SOCIAL-KLD			-0.018**
			(0.008)
D-SOCIAL-KLD ²			0.009***
C'-	0 (10***	0 (14***	(0.001)
Size	0.610^{***}	0.614^{***}	0.498^{***}
II I. J.	(0.008)	(0.008)	(0.012)
n-maex	-0.214	-0.241	-0.088
Current Datio	(0.219)	(0.219)	(0.318) 0.075***
Current-Ratio	-0.070	-0.009	-0.0/3
Dessively last and Laster to Total Access	(0.00/)	(0.007)	(0.009)
Receivables-and Inventory-to-Total-Assets	(0.090)	0.118	(0.421^{+++})
Defermine Alexander	(0.080)	(0.080)	(0.110)
Return-on-Assets	(0.03/(0.121))	(0.039)	-0.055
	(0.131)	(0.131)	(0.102)
Loss-in-Current-Year	(0.042)	0.048^{*}	0.069^{*}
Auditon Tomme	(0.028)	(0.028)	(0.037)
Auditor-Tenure	-0.001	-0.001	-0.012^{+++}
	(0.002)	(0.002)	(0.003)
Auditor-Specialization	(0.123)	(0.022)	(0.028)
Lavanaaa	(0.055)	(0.033)	(0.038)
Leverage	(0.027)	(0.050)	(0.058)
Current Assots to Total Assots	(0.030)	(0.030)	(0.038)
Current-Assets-to-Total-Assets	(0.067)	(0.067)	(0.018^{11})
Salas Crowth	(0.007)	(0.007)	(0.081)
Sales-Olowul	(0.031)	-0.033	-0.027
Markat to Dook Value	(0.039)	(0.039)	(0.048)
Market-to-book-value	(0.002)	(0.002)	(0,003)
Extraordinary Items	(0.002) 0.254***	(0.002) 0.254***	0.0037 0.240***
Extraordinary-richts	(0.234)	(0.234)	(0.240)
Busy-Season	(0.020) 0 114***	0.111***	0.022)
Dusy-Season	(0.017)	(0.016)	(0.004)
Auditor-Change	-0 149***	(0.010)	-0.185**
Tuanor-Change	(0.043)	(0.043)	(0.073)
IFRS	0.021	0.019	(0.0751
	(0.021)	(0.040)	
US-GAAP	0 254***	0 256***	
co ornu	(0.071)	(0.070)	
Country dummies	ves	ves	no
Industry dummies	ves	ves	ves
Year dummies	ves	ves	ves
Constant	3.919***	3.706***	8.126***
	(0.328)	(0.315)	(0.432)
R-squared	0.844	0.844	0.680
Observations	12220	12220	1171

 Table 6. Robustness: CSR Performance and Audit Fees with alternative measures of CSR

 Performance

	I and A. Desell	prive Statistics	s (n-3771)	
Variable	Mean	St. Dev.	Min	Max
Rest	0.0660	0.2484	0.0000	1.0000
CSR-Score	0.5004	0.2863	0.0723	0.9768
CSR- Score ²	0.3324	0.3053	0.0052	0.9540
Total-accrual	0.0362	0.2363	-0.7560	0.5962
Drec	-0.0026	0.0358	-0.1482	0.1130
Dinv	-0.0007	0.0310	-0.1381	0.1231
Soft-assets	0.5541	0.2202	0.0294	0.9720
Dcsale	0.0923	0.1830	-0.5279	2.4226
Droa	0.0001	0.0939	-0.5206	0.4635
Issuance	0.0167	0.1282	0.0000	1.0000
Abret	0.0271	0.3698	-0.7870	1.8030

Table 7: Pre-Audit Misstatement Risk Analysis Panel A: Descriptive Statistics (n=3771)

Panel B: Model for Estimating Pre-Audit Misstatement Risk

DV: Rest	Predicted sign	(1)
CSR-Score	+	3.1926***
		(1.1420)
CSR-Score ²	-	-4.2523***
		(1.1843)
Total-accrual	+	0.0226
		(0.2712)
Drec	+	2.2811
		(1.9115)
Dinv	+	4.4871**
		(2.1944)
Soft-assets	+	0.1201
		(0.2970)
Dcsale	+	0.2693
		(0.3542)
Droa	-	-1.5953**
		(0.7020)
Issuance	+	0.8073**
		(0.3906)
Abret	+	0.2223
		(0.1758)
Abret (t-1)	+	0.0450
		(0.1336)
Constant	-	-1.7658***
		(0.2859)
Observations		3771
Pseudo R2		0.0228
Model likelihood ratio Chi-square		41.83
Model p-value		0.0000

* p < 0.10, ** p < 0.05, *** p < 0.01. Panel A presents the descriptive statistics of the variables used in the analysis of the misstatement detection model, while panel B shows the estimation results. The dependent variable Rest equals 1 if the annual report for the current year is subsequently restated, and 0 otherwise. CSR-Score is the average of the ASSET4 benchmarked Social and environmental Performance scores. It takes values between zero and one. Total-accruals is measured as the change in noncash assets (noncash total assets minus total liabilities and preferred stocks) from year t-1 to year t scaled by average total assets. Drec is the change in accounts receivables over total assets. Dinv is the change in inventory over total assets. Soft-assets are the proportion of soft assets (Total assets - Property, Plants and equipment - Cash and Short-Term Investments) to total assets. Dcsale is the percentage change in cash sales (cash sales = sales – the change in inventory). Droa is the change in Return on assets. Issuance is a dummy variable taking the value of 1 if the firm issued debt or equity securities during year t, and zero otherwise. Abret is the annual buy-and-hold stock return minus annual buy-and-hold value weighted NYSE-AMEX-NASD index return. Standard errors are reported in parentheses.